

Received 7 September 2023, accepted 28 September 2023, date of publication 10 October 2023, date of current version 28 November 2023.

Digital Object Identifier 10.1109/ACCESS.2023.3323399

TOPICAL REVIEW

Federated Learning Meets Intelligence Reflection Surface in Drones for Enabling 6G Networks: Challenges and Opportunities

ALEXEY V. SHVETSOV^{®1}, SAEED HAMOOD ALSAMHI^{®2}, AMMAR HAWBANI^{®3}, SANTOSH KUMAR^{®4}, (Member, IEEE), SUMIT SRIVASTAVA⁵, (Graduate Student Member, IEEE), SWETA AGARWAL^{6,7}, NAVIN SINGH RAJPUT^{®8}, (Member, IEEE), AMR A. ALAMMARI^{®2}, AND FARHAN M. A. NASHWAN² ¹Department of Smart Technologies, Moscow Polytechnic University, 107023 Moscow, Russia

²Electrical Engineering Department, Faculty of Engineering, Ibb University, Ibb, Yemen

³School of Computer Science, Shenyang Aerospace University, Shenyang 110136, China

⁴Department of CSE, IIIT Naya Raipur, Raipur, Chhattisgarh 493661, India

⁵Department of Electronics and Communication Engineering, FET, MJP Rohilkhand University, Bareilly, Uttar Pradesh 243006, India

⁶Department of EC, Invertis University, Bareilly, Uttar Pradesh 243123, India

⁷Department of EE, Invertis University, Bareilly, Uttar Pradesh 243123, India

⁸Department of Electronics Engineering, Indian Institute of Technology (Banaras Hindu University) Varanasi, Varanasi, Uttar Pradesh 221005, India

Corresponding author: Farhan M. A. Nashwan (far_nash@hotmail.com)

ABSTRACT The combination of drones and Intelligent Reflecting Surfaces (IRS) have emerged as potential technologies for improving the performance of six Generation (6G) communication networks by proactively modifying wireless communication through smart signal reflection and manoeuvre control. By deploying the IRS on drones, it becomes possible to improve the coverage and reliability of the communication network while reducing energy consumption and costs. Furthermore, integrating IRS with Federated Learning (FL) can further boost the performance of the drone network by enabling collaborative learning among multiple drones, leading to better and more efficient decision-making and holding great promise for enabling 6G communication networks. Therefore, we present a novel framework for FL meets IRS in drones for enabling 6G. In this framework, multiple IRS-equipped drone swarm are deployed to form a distributed wireless network, where FL techniques are used to collaborate with the learning process and optimize the reflection coefficients of each drone-IRS. This allows drone swarm to adapt to changing communication environments and improve the coverage and quality of wireless communication services. Integrating FL and IRS into drones offers several advantages over traditional wireless communication networks, including rapid deployment in emergencies or disasters, improved coverage and quality of communication services, and increased accessibility to remote areas. Finally, we highlight the challenges and opportunities of integrating FL and IRS into drones for researchers interested in drone networks. We also help drive innovation in developing 6G communication networks.

INDEX TERMS 6G, drones, drone swarm, federated learning, IoT, IRS, smart environment.

I. INTRODUCTION

Recently, the fifth Generation (5G) addresses the significant rise of the Internet of Things (IoT) and users by providing

The associate editor coordinating the review of this manuscript and approving it for publication was Chen Chen^(D).

creative communication options. Due to the capability to improve the capacity of mobile infrastructure and expand coverage, drones are expected to play a vital role in enhancing communication reliability of upcoming wireless networks and the attainable spectral efficiency [1]. Additionally, drones were essentially designed to play a crucial part in 27

20

84

85

86

87

88

89

90

91

92

93

94

95

97

98

99

101

103

104

106

108

109

data distribution to IoT devices [2]. Drones must adopt 28 cutting-edge communication paradigms to fulfil the rising 29 need for high data speeds, but this is difficult given their 30 size and power constraints [3]. To support six Generations 31 (6G), it is very difficult to characterize system models and 32 ensure strict Quality of Service (QoS) requirements in such 33 complex and dynamic network environments. 6G wireless 34 networks are expected to provide various services, combining 35 terrestrial, aerial, and space networks for universal coverage. Most earlier efforts have generally concentrated on commu-37 nications between ground devices and ground Base Station 38 (BS) in the context of the tiny data-packet transmission 39 regime. Non-line-of-sight (NLoS) wireless networks on the 40 ground can sometimes handle many mobile devices while 41 upholding strict mURLLC criteria. Drone can enable several 42 enormous access strategies by significantly improving Line-43 of-Sight (LoS) while guaranteeing various QoS criteria. 44 The benefits of deployment capabilities and high mobility 45 inspired this idea. Intelligent Reflecting Surface (IRS) 46 technology-empowered drone network systems, which have 47 recently been introduced to resolve these problems by 48 avoiding obstructions and enhancing connection in drone 49 systems [4]. According to the IRS-assisted drone design, 50 IRSs are in an open network environment to facilitate 51 communication between drones and smart devices or users. 52 By using the IRS to enable multiple LoS connections, which 53 significantly decrease channel attenuation, a blocked NLoS 54 transmission channel may be addressed. 55

The idea of an IRS has recently come to light as a 56 disruptive technology expected to completely transform wire-57 less communications by giving wireless system engineers 58 complete control over the propagation environment while the 59 wireless transmission is in progress [5]. In particular, IRS 60 is a surface that enables the manipulation of the impinging 61 communication signals to accomplish one of the following 62 goals [6]: (i) extending the coverage to a dead zone, (ii) Phys-63 ical layer security, (iii) extensive Device to Device (D2D) 64 communication, and (vi) wireless data and power transfer. 65 IRSs have an advantage in flexible IoT ecosystems' energy 66 efficient because they do not employ active components. 67 As a result, IRS-assisted drone communications can offer IoT 68 networks energy-efficient communications [1]. 69

Drones fly closer to the implicated battery-limited IoT devices to accomplish energy efficiency, which enables IoT 71 devices to transmit at lower power in the uplink, eventually 72 resulting in decreased energy consumption and extended 73 battery lives [7], [8], [9], [10]. Additionally, the employment 74 of IRS-assisted drones to increase network coverage [11] and 75 channel capacity significantly reduces the number of cellular 76 BS, creating greener networks, smarter, and consuming 77 less energy [12]. The possible use of IRS in cellular 78 communications with drones that have weak signals was 79 examined [13]. In such a scenario, IRS are placed on walls 80 and controlled by a Base Station (BS) to direct reflected 81 signals toward drones. By coordinating the reflections, 82

signal strengths are boosted for the drones, enhancing the wireless communication quality. The symbol error rate and outage probability of multi-layer drone-powered wireless communications provided by the IRS [14]. The authors of [4] examined the integration of drones and IRS by illuminating the uses for IRS and the benefits of drones and outlining the benefits of doing so in combination with the wireless network. The drone trajectory, the transmit beamforming at the BS, and the passive beamforming in IRS are jointly optimized [15].

A potential way to deploy over-the-air intelligent reflection and increase wireless coverage is by merging IRS and drones. IRS-drone integrated systems can drastically lower drone energy consumption and increase operating time because of the attractive benefits of passive IRS. Due to drones' relatively high heights and adaptable 3D mobility, IRS-droneintegrated systems are more likely than terrestrial IRS to establish robust LoS linkages with ground equipment. Addi-100 tionally, IRS-drone integrated systems can achieve panoramic full-range reflection, considerably expanding the number of 102 mobile users supported. Compared to terrestrial IRS systems, several new difficulties exist, including durability, stability, and controllability. Particularly, low-complexity IRS-drone-105 integrated 3D wireless channel models are challenging to describe. Additionally, 3D IRS-drone trajectories with user 107 associations must be designed and optimized to enhance system performance. Characterizing the optimization issues for error-rate and delay-bounded QoS is complex due to 110 high-dimensional complexity, evolving environments, and 111 time-varying action spaces, especially when considering 112 massive access applications to support mURLLC. 113

Despite the benefits of drone communication, the com-114 plicated terrain and surroundings may obstruct the Air-115 to-Ground (A2G) channels. Furthermore, the information 116 security of authorized users could not be assured. IRS can 117 be used in drone-assisted A2G networks to address these 118 problems by creating a favourable propagation environment 119 and enhancing the communication quality of intended 120 users. Furthermore, by appropriately configuring the passive 121 beamforming, the IRS may cancel out the undesired signals to 122 reduce interference and stop aggressive eavesdropping. The 123 performance of A2G networks has recently been improved 124 by experiments fusing drones and IRS [16], [17], [18], [19], 125 [21]. To be more specific, the IRS enables the expansion 126 of drone coverage, supporting a variety of QoS demands 127 from consumers. Furthermore, when mounted on a mobile 128 drone rather than a stationary structure, IRS has more 129 deployment flexibility and a larger range of signal reflection. 130 Recently, research integrating the drone with IRS [22], [23] 131 has emerged to enhance the performance of A2G networks. 132 With the aid of IRS, drone coverage may be increased, and 133 therefore, various QoS requirements of users can be met. 134 When mounted on a mobile drone, IRS has more deployment 135 flexibility and a larger range of signal reflection than when 136 put on a permanent structure. Therefore, the use of drones and 137

Federated Learning (FL) is a decentralized ML paradigm 142 where models are trained collaboratively across multiple 143 devices or servers while keeping the raw data localized. 144 FL finds applications in scenarios where data privacy, 145 security, and distributed data sources are critical concerns, 146 such as mobile devices, edge computing, and sectors like 147 healthcare and finance. Furthermore, the authors of [24] 148 created a unique framework of resource allocation and device 149 selection for the FL technique by deploying numerous IRSs 150 in the FL. Moreover, in [25], focused on enhancing the 151 over-the-air FL (AirFL) performance while maintaining QoS 152 restrictions. When an IRS facilitates the transmission from 153 users to the BS in the AirFL system, [26] explored the 154 model aggregation process. However, the main aim of our 155 novel framework is to provide a comprehensive and novel 156 approach to enhancing communication network performance 157 158 by combining the power of FL and IRS in drone technology. The framework addresses the challenges and seizes the 159 opportunities to deploy IRS in drones and integrate it with 160 FL to enable 6G communication networks. The contributions 161 of the framework can be summarized as providing a solution 162 to enhance communication network coverage, capacity, and 163 energy efficiency through the use of FL and IRS in drone 164 technology. 165

166 A. RELATED WORK

167 1) DRONES

Drones represent a transformative technological innova-168 tion with diverse applications across various fields. The 169 drones, controlled remotely or autonomously, have gained 170 prominence for their ability to traverse challenging terrains, 171 collect data from inaccessible locations, and perform a wide 172 range of tasks without human intervention [27], [28], [29], 173 [30], [31], [32], [33], [34], [35], [36], [37], [38], [39]. 174 Drones come equipped with sensors, cameras, and com-175 munication capabilities, enabling them to gather real-time 176 information, capture high-resolution imagery, and facilitate 177 remote sensing operations. The versatility spans agriculture, 178 surveillance, disaster response, environmental monitoring, 179 and entertainment [40], [41], [42], [43]. UAVs offer the 180 potential to revolutionize industries by providing cost-181 effective, efficient, and flexible solutions to address complex 182 challenges that were previously difficult or impossible to 183 tackle. 184

185 2) IRS

The IRS was implemented initially in drone systems [16] and demonstrated to increase data rate significantly; however, secrecy performance and power allocation should have been considered. The authors then presented a secrecy rate maximization challenge using IRS in [44], developing multi-antenna access points for transmit beamforming and 191 the IRS's reflect beamforming. Furthermore, the authors 192 of [45] investigated the secrecy rate maximization issue 193 using IRS with an eavesdropper and a single receiver. 194 When the genuine receivers' channel response was highly 195 associated with that of the eavesdroppers, the IRS was used 196 to offer extra communication lines [46]. In [47], the authors 197 examined the relevance of generated noise in IRS-aided 198 wireless communication networks. In addition, a unique deep 199 reinforcement learning-based secure beamforming technique 200 is provided for the first time in IRS-aided wireless secure 201 communication to obtain the best beamforming policy against 202 eavesdroppers [48]. The first flying IRS was suggested to 203 protect the terrestrial transmission in the availability of an 204 eavesdropper [18], with the IRS phases, user association, tra-205 jectory, and transmit power all optimized together. However, 206 direct communications between the BS and the users were 207 believed to be prevented. Motivated by the advantages of 208 both the drone and the IRS, the authors of [49] proposed a 209 secure IRS-aided drone to support wireless communication 210 situations such as concerts where large crowds and heavy 211 communication traffic are required temporarily. 212

3) IRS IN DRONE

The combination of drones with IRS for sky reflection was 214 researched in [16] and [50]. These pieces may generally 215 be divided into two groups. Terrestrial IRS-assisted drone 216 communications are one, while drone IRS-assisted com-217 munications are the other [16], [22], [51]. To increase the 218 average attainable rate for the ground user, a hybrid drone 219 trajectory and terrestrial IRS passive beamforming design 220 were examined in [16]. By concurrently designing the drone 221 movement, the terrestrial IRS phase shift, and the power 222 allocation strategy, the authors in [51] used the decaying deep 223 Q-network to reduce the energy consumption of ground users. 224 Using passive beamforming at the terrestrial IRS, a drone was 225 developed to aid the terrestrial IRS in reflecting its signals to 226 the BS and improve drone transmission [22]. 227

The transmitter and receiver are on the ground for 228 combining drone and IRS-assisted communications, resulting 229 in A2G channels. Additionally, the drone and IRS location 230 combination defines array response in the LoS passive signal 231 reflections. The combination of drones and IRS was used 232 to increase the worst-case signal-to-noise ratio in a given 233 region [52]. However, the authors did not consider A2G LoS 234 linkages. Neglected were the NLoS connections affected by 235 the combination of drone and IRS location. Furthermore, only 236 single-user beamforming was considered when the worst-237 case SNR was maximized. Maximizing the drone trajectory 238 and the combination of drone and IRS phase shift, transmit 239 power, and user association motivated the authors of [53] to 240 explore secure up-link communications with the combination 241 of drone and IRS assistance. The NLoS A2G links and several 242 antennae at the BS were ignored, and the combination of 243 drone and IRS height was fixed [53]. 244

Due to the considerable far-field double route loss (i.e., 245 the attenuation or loss of a wireless signal's strength as it 246 travels over a long distance through two distinct propagation 247 paths), the combination of drone and IRS placement design 248 is essential for the combination of drone and IRS-assisted 249 communications [54]. In [52], the optimal LoS A2G route 250 was presumed, and the deployment of a combination of 251 drones and IRS was considered. The combination of drone 252 and IRS placement between destination nodes and the fixed 253 source was the main emphasis [50]. However, the random 254 geographical distribution of users must be considered while 255 evaluating performance. Notably, single drone networks 256 focused on previous efforts that combined drones with IRS. 257 The numerous drone-enabled combinations of drone and IRS 258 systems need to be looked into since the swarm network of 259 drones they generate makes passive signal reflections more effective at increasing aperture gain. References [55] and [56] 261 regarded as relay BSs for multi-drone 3D deployment, with 262 the Point-to-Point (P2P) signal transmissions occurring over 263 the G2A or A2G channels. However, transmissions encounter 264 cascading G2A and A2G channels when drones are outfitted 265 with IRS to reflect signals. 266

267 4) FL MEETS IRS IN DRONE

The combination of FL and IRS has been used in several 268 previous works, including [57] and [58]. By carefully 269 choosing users and allocating resources, the developers of 270 [57] reduced the FL loss function and established an explicit 271 link between packet error rate and FL performance. Ni et 272 al. [58] created a unique resource allocation framework and 273 selected a smart device for the FL system by deploying 274 numerous IRSs in the FL system. 275

276 **B. MOTIVATION AND CONTRIBUTIONS**

Integrating FL and IRS in drones can support the demands 277 of emerging applications in the next generation of wireless 278 communication, such as autonomous drones, aerial pho-279 tography and delivery services, environmental monitoring, 280 disaster response, and infrastructure inspection. These are 281 just a few examples of the potential applications of FL and 282 IRS for drones. Integrating these technologies can bring new 283 capabilities and opportunities to drones' next generation of 284 wireless communication. The integration of FL and IRS in 285 drones is motivated by the need to improve communication 286 performance, security, and efficiency in the next generation 287 of wireless communication for drones. 288

The motivation behind combining FL and IRS in drones 289 to enable 6G networks is to address some critical challenges 290 and opportunities in the next generation of wireless com-291 munication for drones. With the increasing use of drones 292 for various applications, there is a growing demand for new 293 solutions to handle the increasing traffic and improve com-294 munication performance. FL and the IRS have the potential 295 to address these challenges by providing a decentralized and 296 efficient way of training Machine Learning (ML) models 297

and dynamically controlling the wireless environment for 298 drones. By combining FL and IRS for drones, 6G networks 299 can leverage the benefits of both technologies to create a 300 more intelligent and efficient communication infrastructure 301 for drones. Using FL can allow for the decentralized and 302 distributed training of ML models, taking advantage of the 303 large and diverse data generated by drones in the network. 304 In addition, the use of IRS can improve communication 305 performance and security by dynamically controlling the 306 wireless environment and reducing interference. In this 307 framework, we introduce FL meets IRS in drones to enable 308 6G networks and discuss the state-of-the-art research and 309 development in this area. In addition, we summarize the 310 current work and identify the key challenges, opportunities, 311 and future directions for this integration. The contributions of 312 the summaries are as follows: 313

- 1) We provide an overview of the current status of the 314 research and development of FL and IRS in drones 315 for enabling 6G networks. The combination of FL and 316 IRS in drone technology to enhance 6G communication 317 networks is highlighted. Then, we identify the key 318 technical and implementation issues that must be 319 addressed to deploy the framework and achieve its 320 potential benefits successfully. 321
- 2) We introduce a novel framework that can overcome the challenges of combining FL and IRS in drones to enable 6G networks, providing insights into the design and development of FL and IRS integration in drones.
- We identify the critical challenges, opportunities and future trends in integrating FL and IRS in drones for enabling 6G networks.

C. PAPER STRUCTURE

The paper structure is shown as shown in Fig.1. The rest of the paper is organised as follows. Section II discusses the preliminaries, while Section III introduces IRS-Drones. Section IV presents a framework for FL meets IRS in drones to support 6G. Section V addresses the challenges and future trends. Finally, the conclusion of this paper is given in Section VI.

II. PRELIMINARIES

A. 6G

6G is the sixth generation of wireless communication 339 technology expected to succeed in the current 5G networks. 340 It is envisioned as a transformative technology that will 341 revolutionize the way we communicate and interact with 342 technology. 6G networks are expected to be faster, more 343 efficient, and more reliable than the existing wireless 344 networks, with the ability to support many devices and 345 applications. Some of the key features and capabilities that 346 are expected to define 6G networks include higher data rates, 347 ultra-low latency, greater energy efficiency, higher spectrum 348 efficiency, improved security and privacy, and the ability to 349

322

323

324

325

329

337

support new emerging applications and technologies such as
 augmented reality, virtual reality, and the IoT [59].

One of the main challenges in developing 6G networks 352 is the need for a new wireless spectrum that can support 353 the high-speed and high-frequency transmissions required 354 by 6G networks. Another challenge is the need for new 355 and innovative network architectures and technologies to 356 support the massive amounts of data and traffic generated 357 by 6G devices and applications. Despite these challenges, 358 the potential benefits of 6G are immense, and it is expected 359 to drive innovation and growth in various industries such as 360 healthcare, transportation, manufacturing, and entertainment. 361 However, the development of 6G networks is still in its early 362 stages, and it is expected to take several years before the 363 technology is commercially available. 364

365 **B. FL**

FL is a relatively new ML paradigm that allows multiple 366 parties to train a model collaboratively without sharing their 367 data. The basic mechanism of FL involves the distribution 368 of the ML model to multiple devices or clients, each of 369 which trains the model using its local data. The model 370 updates are then sent to a central server for aggregation, 371 and the aggregated model is sent back to the devices for 372 further training. This process is repeated multiple times 373 until the model achieves the desired accuracy. The pros of 374 FL include improved data privacy and security since the 375 data never leaves the device, the ability to train models 376 on decentralized data, and the potential for faster training 377 times since the data does not need to be transmitted to a 378 central server. However, the cons of FL include increased 379 complexity in model optimization, potential communication 380 latency and bandwidth limitations, and the need for careful 381 design and implementation to ensure privacy and security. 382 Table 1 illustrates the cons and pros of FL based on different 383 parameters. 384

385 1) LOCAL MODEL TRAINING

The local model training step can involve using gradientbased optimization algorithms, such as Stochastic Gradient Descent (SGD) or its variants.

2) GRADIENT COMPUTATION

For a model with weights W and biases &b&, the gradient of the loss function L with respect to W and b can be computed as: 390 391 392 392

$$\nabla L(W, b) = \frac{\partial L}{\partial W}, \frac{\partial L}{\partial b}$$
 (1) 393

389

394

395

396

397

398

399

400

403

404

405

406

410

This equation represents the gradient of the loss function L concerning the weights W and biases b of a model. The notation $\partial L/\partial W$ and $\partial L/\partial b$ indicate the partial derivatives of the loss function concerning W and b, respectively.

3) WEIGHT AND BIAS UPDATES

The weights and biases can be updated using the computed gradients:

$$W' = W - \eta \nabla L(W, b) \tag{2}$$

$$b' = b - \eta \nabla L(W, b) \tag{3}$$

where η is the learning rate that controls the step size of the updates, and $\nabla L(W, b)$ is the gradient of the loss function *L* with respect to *W* and *b*.

4) FL AGGREGATION

The FL aggregation process can involve weighted averaging or other aggregation techniques, which can be represented mathematically.

5) WEIGHTED AVERAGING

For the model updates received from each drone, the global 411 model can be computed as a weighted average of the local 412 model updates, where the weights represent the contribution 413 of each local model update: 414

$$Global_{model} = \frac{\sum_{i}^{N} (w_i * Local_{model_i})}{\sum_{i}^{N} w_i}$$
(4) 41

where w_i represents the weight assigned to the *i*-th local 416 model update, which is determined based on various factors 417 such as the performance or reliability of the local model. 418



FIGURE 1. Paper structure.

Parameter	Pros	Cons
Data Privacy	Data is kept on devices, reducing the risk of data breaches	Increased complexity in implementing privacy-preserving
		techniques
	Protects sensitive personal data	Possible vulnerabilities
Decentralized Data	Allows for training on data that is distributed across different	Increased communication overhead between devices
	devices	
	Improving model generalization and accuracy	Possible limitations in network bandwidth and latency
Speed	Reduced training time by training models in parallel on differ-	Increased complexity in model optimization
	ent devices	
	Reduces the need for transmitting large amounts of data to a	Limited by the computing resources available on individual
	central server	devices
Security	Reduces the risk of a single point of failure	Increased complexity in securing communications between
		devices
	Improving security by not transmitting data to a central server	Possible vulnerabilities
Robustness	Continuing to function even if some devices drop out or fail	Increased complexity in aggregating model updates from mul-
		tiple devices
	Adapting to changes in the network environment	Possible reduction in accuracy

TABLE 1. Cons and pros of FL for support framework.

REFLECTION COEFFICIENT OPTIMIZATION 419

The reflection coefficient optimization can involve mathe-420 matical optimization techniques, such as convex optimization 421 or heuristic algorithms, depending on the specific approach 422 used in the proposed framework. 423

C. IRS 424

Intelligent Reflecting Surfaces (IRS) are a new and innovative 425 technology being actively researched and developed for 426 future wireless communication systems. An IRS consists 427 of many passive reflecting elements, each of which can be 428 controlled individually to manipulate the wireless signal that 429 passes through it. By adjusting the phase and amplitude of 430 the reflected signal, an IRS can enhance signal strength, 431 improve signal quality, and reduce interference and noise. 432 IRS technology has several advantages over traditional 433 wireless communication technologies. First, it is much more 434 energy-efficient, as it only requires passive elements to 435 manipulate the signal. Second, it can provide higher data 436 rates and improved coverage, especially in environments with 437 obstacles and interference. Third, it can be easily integrated 438 into existing wireless communication systems and coexist 439 with other wireless technologies. 440

IRS technology has many potential applications, including 441 indoor and outdoor wireless communication, Internet of 442 Things (IoT) networks, 5G and beyond 5G cellular networks, 443 and satellite communication systems. IRS can also be used 444 in conjunction with other technologies, such as drones and 445 artificial intelligence, to enhance the performance of wireless 446 communication systems further. Despite its many advantages, 447 some challenges still need to be addressed before the IRS 448 can be fully integrated into wireless communication systems. 449 These challenges include the design and optimization of IRS 450 structures, the development of efficient control algorithms, 451 and the development of cost-effective and scalable manufac-452 turing processes. 453

D. FL MEETS IRS 454

Integrating IRS and FL involves combining the benefits of 455 both technologies to improve the efficiency and effectiveness 456

of wireless communications and machine learning. There are several critical steps involved in this integration:

Incorporating IRS into the wireless communication The first step is to incorporate IRS into the system: 460 wireless communication system by placing the reflecting 461 surfaces strategically to optimize signal strength and reduce 462 interference [60], [61]. Therefore, it involves modelling the 463 wireless propagation environment and using optimization 464 algorithms to determine the optimal placement of the 465 reflecting surfaces. 466

Using FL to train the ML models: The second step is to use FL to train ML models that can optimize the use of the wireless communication system. This can involve distributing the training process across multiple devices and aggregating the results to create a global model 471 that can be used to optimize the wireless communication 472 system.

Combining the IRS and FL components: The final step 474 is to combine the IRS and FL components to optimize the 475 wireless communication system using ML models. This can 476 involve using the global model to control the phase shifts of 477 the reflecting surfaces to optimize signal strength and reduce 478 interference. The ML models can also optimize allocating 479 resources such as bandwidth and power to different users 480 based on their specific needs. 481

Summary: The Integrating IRS and FL improve wireless communication systems' efficiency and effective-483 ness by optimizing resource use and reducing interfer-484 ence. The combination of these technologies can also 485 enable new applications such as smart transportation sys-486 tems, remote healthcare, and industrial automation. How-487 ever, careful consideration must be given to data privacy 488 and security issues to ensure this integration's benefits 489 are realized without compromising individual rights and 490 freedoms. 491

Figure 2 shows the steps involved in the FL algorithm for 492 optimizing reflection coefficients, including data collection 493 from the communication environment, local model training at 494 each IRS, local model updates, FL aggregation, global model 495 update, and model distribution. 496

458 459

467

468

469

470

473

482



FIGURE 2. FL algorithm for optimizing reflection coefficients.

497 E. FL IN UAV-ENABLED NETWORKS

FL in UAV-enabled networks involves multiple UAVs 498 collaboratively improving machine learning models while 499 keeping data local [62], [63]. Each UAV collects data from 500 its environment and trains its model on the device. Instead of 501 sharing raw data, UAVs transmit model updates to a central 502 coordinator, aggregating these updates to create an improved 503 global model. The model is then returned to UAVs for 504 integration into their local models. The approach preserves 505 data privacy, reduces communication overhead, and adapts 506 to diverse environments, aligning well with UAV networks' 507 distributed nature and limited communication capabilities. 508

Compared to traditional centralized learning, FL offers 509 distinct advantages. It safeguards sensitive data by avoiding 510 data transmission and ensuring privacy. Communication 511 efficiency is enhanced since only model updates are shared, 512 reducing bandwidth demands. The decentralized nature 513 improves resilience and scalability, accommodating the 514 dynamic UAV network. Moreover, FL allows UAVs to adapt 515 models locally to context-specific conditions, resulting in 516 more accurate insights. The energy-efficient approach caters 517 to UAVs' limited power resources and facilitates real-time 518 decision-making. Table.2 provides a simplified comparison 519

Algorithm 1 Processing Flowchart

1: Initialization: The ensemble process initialises the IRS reflection coefficients to a random value and the global model weights to zero.

2: Local model training: Each drone in the swarm trains a local model on its data using FL. The local model is updated using the gradient descent algorithm with the learning rate. The loss function used for optimization is a weighted average of the mean squared error (MSE) of each drone's data, with weights determined by their performance or reliability.

3: Model aggregation: The local models are sent to the central controller, aggregating them using a weighted average. The weights assigned to each local model are determined by their performance or reliability.

4: IRS reflection coefficient optimization: The central controller then uses the aggregated model to optimize the IRS reflection coefficients. This is done by minimizing the loss function using the gradient descent algorithm with the learning rate.

5: Global model update: The central controller updates the global model weights using the aggregated local models and the optimized IRS reflection coefficients. This is done using the formula

$$W' = W - *L(W), \tag{5}$$

where W is the global model weights and L(W), is the gradient of the loss function concerning W.

6: Repeat: The ensemble process is repeated for multiple iterations until convergence is achieved.

comparison table between FL and traditional centralized 520 learning in the context of UAV-enabled networks 521

1) WILDLIFE CONSERVATION AND MONITORING

UAVs equipped with cameras and sensors can be deployed 523 to monitor wildlife habitats. Each UAV collects data on 524 animal behavior, habitat conditions, and ecological changes. 525 Federated Learning enables UAVs to collaboratively train 526 models to identify animal species, track migration patterns, 527 and detect unusual behavior without sharing sensitive loca-528 tion data. This approach ensures data privacy, as raw images 529 and locations remain localized while contributing to a global 530 model that improves species protection and conservation 531 efforts. 532

2) DISASTER RESPONSE AND RECOVERY

In disaster-stricken areas, UAVs can rapidly assess damages and identify survivors. Federated Learning empowers UAVs to develop models for damage detection, survivor identification, and emergency resource allocation. Each UAV processes images and sensor data locally to enhance situational awareness while respecting individuals' privacy. These models can quickly adapt to changing conditions and

522

Items	FL	Centralized Learning
Data Privacy	Preserves privacy by keeping data on UAVs	Raises privacy concerns due to central data storage
Communication	Minimizes communication by transmitting only model up-	Demands substantial data transmission to a central server
Efficiency	dates	
Decentralization	Aligns with the UAV network's distributed structure	Centralizes data, potentially leading to single points of failure
Adaptability	Adapts models to local UAV environments	Lacks adaptability to diverse contexts without continuous data
		updates
Scalability	Scales naturally as more UAVs join	face scalability issues as data volume grows
Energy Efficiency	Energy-efficient with local training and lightweight updates	May strain energy resources due to data transmission
Resilience	Enhances network resilience by removing central dependency	Vulnerability to server failures impacting the entire network
Real-time Decision-	Supports real-time updates due to lightweight communication	Might experience latency in model updates due to data trans-
making		fers

TABLE 2. Comparison between FL and traditional centralized learning in the context of UAV-enabled networks.

improve disaster response efficiency by facilitating accurate 541 decision-making without centralized data accumulation. 542

PRECISION AGRICULTURE 543

UAVs play a vital role in precision agriculture by monitoring 544 crop health, irrigation needs, and pest infestations. Federated 545 Learning enables UAVs to train models that provide real-time 546 insights to farmers. Each UAV collects data on crop 547 conditions and environmental factors, allowing models to 548 optimize irrigation, pest control, and crop yields. Privacy is 549 preserved since raw data, such as field images, remains on 550 UAVs, ensuring sensitive farming practices stay confidential. 551

4) TRAFFIC MONITORING AND CONTROL 552

UAVs can manage traffic by monitoring road conditions, 553 congestion, and accidents. Federated Learning allows UAVs 554 to develop models that predict traffic patterns, optimize traffic 555 flow, and detect road hazards. Each UAV gathers traffic data 556 from its vantage point while avoiding data centralization. This 557 approach enhances urban mobility while addressing privacy 558 and data security concerns. 559

5) ENVIRONMENTAL SURVEILLANCE 560

UAVs are essential for monitoring environmental changes 561 such as pollution, deforestation, and climate-related events. 562 Federated Learning enables UAVs to build models that 563 analyze air quality, vegetation health, and natural disaster 564 impacts. Each UAV gathers data specific to its loca-565 tion, contributing to global insights while preserving data 566 confidentiality. This approach enhances our understanding 567 of environmental trends without compromising sensitive 568 geographic information. 569

III. IRS-DRONES 570

IRS and drones are two emerging technologies that have 571 gained significant attention recently. For example, IRS 572 is a passive and intelligent radio environment that can 573 manipulate radio waves through many low-cost, low-power, 574 and controllable phase shifters. On the other hand, drones can 575 fly autonomously or be remotely controlled. Integrating IRS 576 and drones presents a promising solution for various com-577 munication and networking applications, such as enhancing 578

network coverage [64], increasing network capacity [64], 579 and reducing energy consumption [65], [66]. In particular, 580 the deployment of IRS in drones can allow on-demand radio 581 environment control and dynamic adjustments to commu-582 nication links, leading to improved network performance. 583 Additionally, by integrating IRS into drone technology, it is 584 possible to increase the network coverage area and provide 585 communication support to remote or difficult-to-reach areas. 586 The IRS-drone integration represents a novel approach to 587 enhancing communication networks and is expected to 588 impact the development of 6G communication networks 589 significantly.

A smart IRS may be able to intellectually manipulate 591 the FL-aware IRS Task Scheduling (FL-IRSTS) approach 592 to extend flag concentration attained at the objective [67]. 593 This is frequently contrasted to earlier practices that improved 594 distant communications by modifying the recipient or sender. 595 An IRS is composed of many IRS, each of which is 596 capable of speaking to the occurrence flag at various points. In IRS-assisted communications, the remote connection 598 from the source to the IRS optimized the objective. Such 599 a communication strategy is invaluable when there is no 600 clear LoS between the origin and destination or a weak 601 distant channel due to borders or poor natural circumstances 602 [68]. Many experts in remote communications anticipate that 603 the IRS will significantly enhance 6G systems by effectively 604 tailoring remote communication scenarios. 605

The IRS is known as metasurfaces [6] and is an emerging 606 technology that can help wireless data transmission networks 607 work more efficiently. IRS' major goal is to increase the 608 quality of wireless communications by raising their total 609 energy by managing the propagation medium. Because of 610 its tremendous influence on energy and spectral efficiency, 611 IRS technology is predicted to play a vital role in enhancing 612 6G networks. The IRS consists of passive antenna elements 613 capable of adjusting the phase of wireless signals before 614 reflecting them to the intended targets. To optimize trans-615 mission efficiency, multiple reflectors are employed for a given target, each with chosen phase shifts that align the 617 reflected signals coherently in the channel. The wireless 618 signal propagation environment is intentionally altered by 610 manipulating numerous small reflecting elements, making 620 IRS a potential candidate for improving diverse aspects 621

590

of forthcoming wireless communication. While individual 622 nodes within the IRS gain empowerment, the central node's 623 involvement remains vital for decision-making and data 624 learning during the IRS process. IRS is a new reflecting 625 radio technology that has gotten much buzz recently [69]. 626 Furthermore, IRS has a 2D artificial structure such as an 627 array of reflective elements that can be strategically con-628 figured to control signal propagation and enhance wireless 629 communication, with many passive reflective elements whose 630 electromagnetic properties, such as reflection, scattering, and 631 refraction, can be controlled electronically and independently 632 in real-time by applying various control signals. 633

The reflected signal's direction may be precisely regu-634 lated for a particular receiver [70], allowing for a fully 635 programmable radio environment. The ability to program 636 the radio space in such a way has enormous potential 637 for wireless networks. For example, IRS can enhance the 638 received SNR and improve the throughput and coverage of 639 wireless communication networks by reflecting signals from 640 a transmitter (TX) to the receiver (TR) [71]. Furthermore, 641 642 the SNR improves significantly, allowing high modulation orders to be used to enhance spectral efficiency. Holographic 643 MIMO surfaces, which may shape electromagnetic waves to 644 meet specific goals, has recently received a lot of interest [72]. 645 Similarly, by regulating the propagation of radio waves in a 646 specified region of interest, IRS may be used to cancel or 647 decrease hazardous wireless interference [73]. The authors 648 of [74] consider merging IRS with Simultaneous Wireless 649 Information and Power Transmission (SWIPT). In [75], the 650 authors proposed a solution for joint optimization issues to 651 652 adjust transmit and reflection beamforming to achieve the minimal weighted received signal strength to interference 653 plus noise ratio at users with transmit power limits. IRS 654 is utilized in [76] to improve the coverage of network by 655 using multi-hop transmission with numerous IRS panels 656 and deep reinforcement learning to create the beamforming 657 matrices. 658

IRS may change the amplitude and phase of incident 659 signals using many low-cost passive reflecting pieces, 660 making it a viable technology for reconfiguring propagation 661 662 conditions and improving network performance [6]. IRS consumes significantly less energy than current systems 663 like active relay and backscattering communication, and it 664 can be installed on building facades, walls, and ceilings. 665 IRS has recently been studied in terrestrial networks to 666 improve energy efficiency, capacity, and security [44], [77], 667 and [78]. The phase shifts of reflecting components can 668 be adjusted in conjunction with the transmission control of 669 transceivers in various network configurations to achieve 670 various communication goals. IRS has recently been touted 671 as a possible method for quicker and more reliable data 672 transfer [79]. In the last few months, a slew of novel research 673 has been committed to the IRS due to its controllability, 674 energy efficiency, and environmental adaptability. Three 675 more features of the IRS are critical to implementing the 676 utilities mentioned above and their widespread deployment. 677

130868

First, IRS comprises energy-efficient, cost-effective passive devices such as printed dipoles [78]. Second, IRS can be made in a high-density configuration [80]. Third, the IRS may be controlled electronically with a rapid switching rate between states, allowing real-time reconfiguration of reflected waves.

683

A. IRS-ENABLE DRONE

The use of IRS is a potential technique for 6G networks. The 684 IRS's cutting-edge technology controls wireless propagation 685 and directs the signal in a specific direction using passive 686 reflecting components. To improve spectrum efficiency, some 687 contemporary literary works have suggested fitting drones 688 with IRSs and using the IRS to reflect the signal in the 689 direction of drones flying BSs. However, Figure 3 depicts 690 these two various IRS-drone coupling scenarios: (i) drones 691 for IRSs, where the drones carrying the reflective surfaces 692 can function as a passive relay in both downlink and uplink communications between ground users and terrestrial BSs, 694 and (ii) IRSs for drones, where IRS-equipped buildings aid 695 the drone's communication. 696



FIGURE 3. Empowering Drone Communication Networks with Drone-IRS and IRS-Enhanced Infrastructure.

The authors of [18] presented an effective deployment of 697 IRS-equipped drones to service a mobile user who does not 698 reach LoS with the BS in supporting mmWave technology. 699 IRS collects energy from the drone and reflects mmWave 700 signals. The deep Q network is utilized to define the location 701 of the drone, and the IRS reflecting parameters to maximize 702 the capacity of the downlink. The same authors discussed 703 using IRS-equipped drones for numerous users to get through 704 the bottleneck caused by the buildings [18]. The position 705 of the drone, the BS precoding matrix, and the reflection 706 parameters are optimized using distributional reinforcement 707 learning. In contrast to [18], the study [51] focused on 708 reducing drone energy usage rather than considering energy 709 harvesting. The position, IRS phase shift, and power dis-710 tribution to mobile users are all optimized in the research 711 on increasing drone-IRS's service quality. In addition, 712 a non-orthogonal multiple access approach is employed to 713 increase downlink communication efficiency. Compared to 714

the traditional Q-learning technique, the degrading deep Q 715 network converges and prevents oscillations when used to 716 alter the drone's location dynamically. 717

The authors of [82] investigated the application of the 718 drone-IR as a passive relay for transmitting smart devices 719 to BS from the viewpoint of uplink. To reduce the average 720 age of the information, deep reinforcement learning is used 721 to optimize the location of the drone-IRS, the transmission 722 timing, and the phase shift. In contrast to the works mentioned 723 above, which focus on IRS-equipped drones, the authors 724 of [83] discussed how a drone-IRS is positioned on the 725 top of nearby buildings to enhance the channel condition 726 between the drone and multiple users. Drone trajectory, data 727 throughput, and IRS phase changes are just a few of the 728 variables optimized in this arrangement. The findings showed 729 two distinct types of solutions for the reinforcement learning 730 technique: (i) a continuous action space utilizing the deep 731 deterministic policy gradient (DDPG) technique and (i) a 732 discrete action space based on a deep Q network. 733

Summary: By integrating IRS into drones, the quality of 734 735 the drone and other network nodes can be improved, leading to better connectivity and more reliable communication. For 736 example, IRS can be used to mitigate the impact of obstacles 737 and reflections, providing clear line-of-sight communication 738 between the drone and the other nodes, reducing communi-739 cation latency, and increasing the communication data rate. 740 IRS can also be used to offload computation tasks from the 741 drone to the edge devices, reducing the computation load 742 on the drone and extending its battery life. This enables the 743 drone to perform more computation-intensive tasks like real-744 time data analysis and decision-making. Additionally, IRS 745 can enhance the security of the drone network by providing 746 a secure communication channel between the drone and the 747 other nodes. IRS dynamically controls the reflection of the 748 signals, enabling the secure transmission of sensitive data 749 without being intercepted by unauthorized nodes. IRS enable 750 drones to achieve enhanced communication and computation 751 capabilities, improving connectivity, reliability, security, and 752 performance. Integrating IRS into drones is a promising 753 direction for future drone networks, enabling more advanced 754 and efficient drone applications. 755

B. HARNESSING IRS FOR ENHANCED DRONE 756 COMMUNICATION NETWORK 757

IRS installations are not confined to indoor and outdoor 758 contexts, but when placed over drones for wireless coverage 759 extension, they considerably boost capacity. IRS-integrated 760 drone-based wireless networks have two types of IRSs: (a) 761 IRS for drone-enabled data communication, where drones 762 gather data from scattered nodes, and (b) IRS for drone-aided 763 ubiquitous coverage, where IRSs are installed in drone 764 networks to increase ubiquitous coverage area, (c) IRS for 765 information transfer and energy in SWIPT networks with 766 drones, (d) IRSs can be placed near customers as a gateway 767 to boost backhaul capacity in cases when drones cannot 768

be deployed near consumers owing to insufficient wireless 769 capacity, (e) IRS for drone-aided secrecy communication, 770 here the IRS may be used to improve security in the 771 drone by weakening eavesdropper communication channels, 772 and (f) IRS for cellular-connected in drone communication 773 networks, where IRS passive beamforming may be improved 774 to enhance drone communication downlink and uplink 775 [64]. However, drone communications may be blocked and 776 eavesdropped because of the enormous hurdles and high 777 node mobility in a wireless network. In this regard, IRS 778 installations can improve the performance of future non-779 terrestrial networks by creating a favourable and controlled 780 wireless environment by managing drones' trajectory. 781

The IRS mounted on buildings can help the drone-based 782 integrated A2G network, where the drone trajectory can be 783 tuned and combined with passive and active beamforming 784 to increase the secrecy rate. The main problem, however, 785 is optimizing the drone's trajectory in conjunction with the 786 IRS passive beamforming. The IRS components' location is 787 a significant aspect of enhancing reflection efficiency. Thus, 788 it must be determined appropriately [4]. Multiple drones are 789 used in recent research [84] to plan the IRS deployment to 790 optimize the average attainable rate. The authors of [19] used 791 a downlink NOMA to optimize the position of the drone-IRSs 792 to increase the user rate while keeping the weak user rate 793 constant. Optimizing the received power at the user is defined by maximizing active beamforming at the drone, passive 795 beamforming at the IRSs, and the drone's trajectory during a certain flight period. The authors developed a semi-definite 797 relaxation iterative technique to improve the IRS phase shifts and transmit beamforming.

Drone-IRS trajectory optimization with passive beamforming to maximize capacity is one of the most critical design elements for IRS deployment. However, the two biggest hurdles to optimizing the drone's trajectories are low power consumption and consistent user communication. To overcome the problem, the authors of [16] propose using IRSs to improve the signal quality of communication between a drone and its users. In addition, the authors of [85] investigated IRS deployment for achieving high gains from the drone-IRS arrangement for user connections. The findings showed that the IRS-assisted cellular system might significantly enhance SINR throughout the area where drone trajectory can be adjusted [13], [14]. Moreover, the authors of [18] presented the influence of phase compensation error on the IRS ergodic capacity aided by drone communications. As a result, successful IRS deployment in the non-terrestrial network can assist in enhancing connection quality and offer flexibility in A2G networks.

Summary: IRS deployment in drones involves integrating 818 IRS components into the drone's hardware and software 819 architecture. The deployment process consists of design, 820 implementation, and testing. The deployment of IRS in 821 drones is a complex process that requires a comprehensive 822 understanding of the IRS technology, the drone hardware 823 and software architecture, and the communication and 824

794

796

798

790

800

801

802

803

804

805

806

807

808

809

810

811

812

813

814

815

816

computation requirements. The deployment process can be 825 optimized using advanced design tools, simulation platforms, 826 and testing frameworks, enabling IRS's efficient and effective 827 drone integration. The deployment of IRS in drones involves 828 integrating IRS components into the drone architecture, 829 facilitating enhanced communication and computation capa-830 bilities. The deployment process requires a comprehensive 831 understanding of IRS technology, drone architecture, and 832 advanced design and testing tools. 833

834 C. DRONE-IRS FOR COMMUNICATION

A novel network architecture, drone-IRS, was developed to 835 expand the service region [86], where the IRS is installed on 836 a drone to implement over-the-drone IRS. Due to the com-837 paratively higher altitude of the drone, drone-IRS is likely to 838 establish more robust LoS linkages with the ground nodes 839 than terrestrial IRS, minimizing the likelihood of blockage. 840 Drone-IRS can achieve 360 panoramic full-range reflections, 841 i.e., one drone IRS may help communicate between any nodes 842 within the coverage area, considerably boosting the number 843 of serviceable users. Furthermore, the drone-IRS coverage 844 area can be expanded further by utilizing its high mobility 845 to move closer to multiple area-separated users sequentially 846 to enhance their communication performance by utilizing 847 short-range LoS channels and minimizing the IRS-reflected 848 link product distance. On the other hand, terrestrial IRS 849 can improve drone-ground communication performance by 850 reducing signal reflections [13], [16]. 851

The literature has also looked at the integration of the 852 IRS with drones. For example, Lu et al. [86] recommended 853 that terrestrial users be served by flying platforms such 854 as balloons or drones outfitted with IRS. Because of the 855 capacity to reposition the IRS to maximize specific system 856 characteristics, such as maximizing the SNR, the reported 857 findings indicate that flying the IRS has an additional degree 858 of freedom. Furthermore, it is demonstrated that, as compared 859 to terrestrial IRS, flying IRS requires fewer parts to produce 860 a given benefit. The IRS was utilized to guide the drone's 861 signal to boost its received signal intensity [13]. The results 862 demonstrate that by optimizing the IRS position and phase of 863 the reflected signals, considerable signal enhancement may 864 be achieved with a limited number of reflectors. A system 865 in which a single drone broadcasts to numerous terrestrial 866 IRS was investigated [23]. The research focused on the best 867 beamforming architecture for the drone, IRS, and trajectory 868 to optimize the received power for ground users. The authors of [21] introduced the downlink of a multi-antenna BS using 870 an IRS-drone platform to connect with a single antenna user. 871

The work assesses the IRS's ability to maximize the system's total energy efficiency by beamforming the optimization vector at BS and the IRS's phase shift matrix. Many optimisation strategies are used under the premise of efficient channel state information. Several additional publications have looked at using an integrated IRS-drone to reduce transmitting power, increase SNR, improve spectral efficiency, or increase the sum rate [16], [50], [87], [88]. However, they did not consider the risk of a mistake or outage or the consequence of an inadequate phase estimate and control method. The capability of IRS-based drone communications with unsatisfactory phase adjustment is assessed [77].

The IRS technology offers a promising yet low-cost 885 solution to this problem since it can simulate significant 886 MIMO gain with active antennas [89]. Therefore, single-887 antenna drone-assisted communications have received a lot of 888 research [90], [91]. Nonetheless, by intelligently modifying 889 its reflection coefficients, an IRS may provide a high passive 890 beamforming gain without needing many antennas on a 891 drone. As a result, one of the critical reasons for this effort 892 is to "recycle" some of the dissipated signals by reflecting 893 them on the targeted consumers. IRS in drone-enabled communication systems can increase the freedom to design 895 a drone's trajectory. For example, if a user is far from the 896 drone but near an IRS, the drone does not need to change 897 trajectory and fly close to the end user to establish strong 898 communication linkages, which takes energy and time in 899 consideration. Instead, an IRS can work with the drone to 900 conduct beamforming on the reflected signals to boost the 901 received signal strength at the remote ground user, allowing 902 for reasonable data throughput. 903

Incorporating an IRS into drone-enabled communication 904 systems presents possibilities and problems in determining its 905 combined trajectory and resource allocation. The composite 906 channel power gain combining the direct link from the drone 907 to ground users and the reflected link through the IRS is 908 a complex function of the drone's trajectory because of 909 the IRS. Furthermore, properly scheduling users for IRS 910 assistance is still unclear, and it is worth our time to 911 investigate. Finally, because broadband communications are 912 widely used in today's cellular networks, the reflected path 913 of IRS results in a frequency- and spatial-selective fading 914 channel, posing a significant challenge for drone trajectory 915 design that has previously been overlooked by works based 916 on frequency-flat channel models [91], [92], [93]. Although 917 [94] developed a multi-carrier channel model for IRS-assisted 918 communications, it is irrelevant to drone communication 919 systems since it ignores the drone's mobility. The authors 920 of [18] outfitted a drone with an IRS to increase the 921 dependability of terrestrial millimetre-wave communication 922 networks. An RL strategy was used to optimize the location 923 of a drone and the IRS reflection coefficients to maximize the 924 system sum rate. 925

Summary: Drones integrated with IRS can significantly 926 enhance communication capabilities compared to traditional 927 drone networks. IRS are smart surfaces that can dynamically 928 control the reflection of radio frequency (RF) signals, 929 enabling the manipulation of the propagation environment 930 and improving communication quality. By integrating IRS 931 into drones, the communication quality between the drone 932 and other network nodes can be improved, leading to better 933 connectivity and more reliable communication. Additionally, 934

985

986

987

988

989

990

991

992

993

994

1006

the IRS can enhance the security of the drone-IRS com-935 munication network by providing a secure communication 936 channel between the drone and the other nodes. IRS can 937 dynamically control the reflection of the signals, enabling 938 the secure transmission of sensitive data without being 939 intercepted by unauthorized nodes. As a result, drone-940 IRS communication networks can significantly enhance 941 communication capabilities, providing better connectivity, 942 reliability, security, and performance. Integrating IRS into 943 drones is a promising direction for future drone networks, 944 enabling more advanced and efficient drone applications. 945

946 D. APPLICATIONS OF IRS-DRONE

Combining IRSs and drones can be advantageous in various
 communications and networking applications. This section
 discusses how combining both technologies may affect
 coverage, interference, security, and SWIPT.

951 1) COVERAGE

A drone offers this capacity by carrying the Intelligent Omni 952 Surface (IOS) below it and flying at an appropriate altitude 953 to produce reflecting RF surfaces when necessary. IRS-954 assisted drone technology increases the range of incident 955 signal SNR at the periphery of a BS coverage area. As shown 956 in Figure 4, the initial cell coverage is effectively extended 957 in the required direction by optimizing the drone trajectory 958 and phase shift vectors to reach the intended end users, 959 whether static or mobile. Mahmoud et al. [95] introduced the 960 use of IRS in drone-powered communications networks to 961 increase coverage and boost dependability regarding spectral 962 efficiency while considering the IoT paradigm. The research 963 investigated the ergodic capacity and outage probability after 964 first deriving tractable analytical formulations. 965



FIGURE 4. Drone-IRS coverage areas.

The deployment of IRS on drones can significantly enhance the coverage of wireless communication networks. With IRS, the signal quality can be improved by reflecting the radio waves in a desired direction, reducing the impact of

966

967

968

969

obstacles and reflections, and increasing the coverage range. 970 IRS on drones can also provide an alternative to traditional 971 wireless communication networks, such as cellular networks. 972 In areas with limited or no network coverage, IRS-equipped 973 drones can be deployed to create a communication network 974 that provides coverage in these areas. In addition, IRS drones 975 can provide temporary coverage in emergency response 976 situations, such as disaster zones, urban canyons, and remote 977 locations. IRS-equipped drones can be deployed quickly to 978 provide communication coverage in these areas, improving 979 the efficiency and reliability of emergency response efforts. 980 The integration of IRS into drones has the potential to 981 revolutionize wireless communication networks, providing 982 improved coverage, quality, and reliability. 983

2) CAPACITY

In contrast to half-duplex mode, the IRS typically operates in full-duplex mode, increasing spectral efficiency. IRS is passive, in which any antenna self-interference and noise amplification are eliminated by its relaying technique, resulting in lower power consumption and less computation than active full-duplex relays. Under the premise of knowing CSI at the IRS controller, interference cancellation can be accomplished by adjusting the phase shifts of particular IRS parts to invert the interference signal and eliminate or lessen it.

Additionally, by maximizing the phase shifts of the antenna 995 components, the IRS may work with the drone to create 996 rich dispersion of LoS connections for many ground end 997 users. The IRS properties and the LoS capabilities made 998 possible by the drone will cause the spectral efficiency to 999 outperform the other capacity-increasing methods. Where 1000 available, multiple drones paired with static IRS can be used 1001 to provide scalability. It is important to note that aerial-IRS 1002 can enable LoS connections with strong channel quality (high 1003 SNR) to increase spectral efficiency by implementing spatial 1004 multiplexing and/or multi-user MIMO. 1005

3) MASSIVE MULTIPLE ACCESS

The difficulties of large access may be successfully overcome 1007 by combining IRS technology with the drone's dynamics and 1008 improving the IRS phase shift vectors to increase system 1009 capacity [95]. Communications systems supported by the IRS 1010 direct indoor wireless channels in favour of users with distinct 1011 needs from typical users. However, the practicality of outdoor 1012 Virtual Reality (VR) applications using IRS-assisted drone 1013 communications systems may increase. 1014

Three main issues are anticipated to affect indoor and outdoor VR users: multi-link communications, energy consumption brought on by massive data transfers, and interference from nearby VR equipment [96]. When drones are an essential component of IRS communication systems, these problems may be successfully solved. In [16], the authors use this technique to optimize the IRS jointly with

the drone height and trajectory phase shift vectors to improve 1022 coverage and capacity and enable widespread connection. 1023

4) SPECTRUM SHARING 1024

The energy consumption of drone systems is crucial to 1025 the system's long-term performance. Thus, IRS-assisted 1026 drone systems can use spectrum sharing to increase system 1027 capacity in hot locations. The viability and benefits of IRS 1028 providing spectrum sharing for indoor smart environments 1029 have been demonstrated [73]. In this case, the capacity 1030 is maximized by allowing the multi-user to access the 1031 shared spectrum. In contrast, user interference is managed 1032 by IRS optimization of phase shifters. The authors of 1033 [97] introduced the IRS-aided spectrum-sharing method to 1034 boost secondary users' capacity while ensuring primary 1035 users' QoS by channel diagonalization and phase shift 1036 optimization. The natural continuation of these studies is 1037 spectrum sharing facilitated by IRS-assisted drones, where 1038 the parameters characterizing the drone mechanics will play a 1039 significant part in real-time wireless networking performance 1040 optimization under dynamic user settings. A cooperative 1041 multiple-task reallocation problem with target precedence 1042 constraints for heterogeneous UAVs was addressed [98], 1043 [99], utilizing a combination of fuzzy C-means clustering 1044 and ant colony optimization algorithms. To increase the 1045 system capacity with IRS installed on drones, it is crucial to 1046 understand how altitude, latitude, and longitude coordinates 1047 affect the performance of the IRS phase shifters. 1048

5) SECURITY 1049

Drones have been suggested to enhance terrestrial cellular 1050 networks' Physical Layer Security (PLS). The dominating 1051 LoS connections that may be made between an aerial and 1052 ground node make this possible. PLS assistance comes in 1053 various forms, from drones. Drones, for example, can serve 1054 as an AR between authorized users to maximize transmission 1055 power and reduce the data rate for eavesdroppers. Drones 1056 can also be used as friendly jammers to broadcast powerful 1057 artificial noise that can reach potential attackers and shield 1058 the data and privacy of genuine users. In normal wireless 1059 contexts, the aforementioned drone functions enhancing PLS 1060 have shown tremendous promise [20]. The development 1061 of wireless threats and assaults, on the other hand, has 1062 led to the creation of complex and challenging situations 1063 that can impair the functioning of wireless networks even 1064 when the suggested safety measures are used. For example, 1065 an eavesdropper can carefully place himself to acquire a high 1066 SNR, possibly greater than the destination node. To combat 1067 cunning attackers, IRSs placed on drones can be used. Prior 1068 studies have demonstrated that when the distance between 1069 peers reduces, the secrecy rate among legal users rises. 1070 As a result, the free movement paradigm of drones reduces 1071 the transmission source's distance from the target user. The 1072 IRS phase shifts can then be adjusted so that the original 1073 signal and the reflected signal at the authorized user combine 1074

positively to increase the SNR. To reduce the received SNR in 1075 particular areas and reduce the likelihood of eavesdropping, 1076 some of the IRS reflecting units, on the other hand, can use 1077 various phase shifts to generate a destructive reflected signal, 1078 as shown in Figure 4. 1079

1080

1111

E. IRS FOR ENABLING DRONE SWARM

In this section, we discuss the importance of IRS for 1081 enabling drone swarm. There are various uses for drone 1082 swarm in wireless networks, including traffic offloading in 1083 hotspots, surveillance, IoT networks, drone swarm networks 1084 in catastrophes, Vehicle-to-Everything (V2X) communica-1085 tions helped by drones, and the creation of smart cities. drone 1086 swarm cooperating to complete a task better than one drone. 1087 Drones are frequently used for military purposes, but interest 1088 in their civilian applications has recently grown. By creating 1089 optimum reflector coefficients, the passively reflecting IRS 1090 may amplify signals at receivers and lessen interference. IRS 1091 has minimal power consumption without active transmitters 1092 due to its passive nature. Therefore, despite the ground 1093 IRS deployment, the drone-enabled aerial IRS may be 1094 effectively deployed and offer panoramic reflections for 1095 ground communications [9]. 1096

The swarm of drone-IRS systems provides the following 1097 benefits over drone-IRS: (i) raising the drone count to 1098 enhance aperture gain; (ii) ensuring drone flight stability and 1099 adaptability by allowing each drone to have a moderate-1100 sized IRS, particularly in adverse weather or air turbulence; 1101 (iii) offering a rich scattering environment with various 1102 drones' placements, which facilitates spatial multiplexing 1103 for a large number of users; (iv) the production cost of 1104 IRS can be lowered, and the flight time of the drone can 1105 be increased thanks to the smaller IRS size of each drone. 1106 In [7], the authors introduced the trade-offs between energy 1107 consumption, latency, and dependability in drone swarm 1108 networks with random network coding. 1109

The benefits of a swarm of drone-enabled IRSs are 1110 enumerated as follows:

- 1) A more significant gain in the aperture It has been 1112 demonstrated in earlier publications [78] that IRS 1113 obtains a power gain via reflection-beamforming and 1114 captures a power gain by gathering the incoming signal 1115 energy. 1116
- 2) Collaborative communication: Future wireless net-1117 works will allow a drone swarm to operate together in 1118 civic and military applications, including surveillance, 1119 video streaming, and combat monitoring. As a result, 1120 drones with various features (such as drone BS, drone, 1121 and drone user) may work together to offer reliable 1122 wireless communication. 1123
- 3) Ensuring flight path Stability and adaptability: A drone 1124 has several reflected components, and it is challenging 1125 to provide flying flexibility and stability, especially 1126 in adverse weather or turbulent air. Furthermore, 1127 an additional payload would need more energy from 1128

1188

1199

1200

the drone, which would shorten its lifespan, given the limited battery capacity. To guarantee flexible mobility and a longer lifespan, each drone can carry fewer reflected components when more drones are present. Additionally, drone swarm follows moving automobiles in vehicular networks, supporting reliable communication with their adaptable mobility.

4) Reduced IRS costs: A large IRS has more expensive manufacturing since it has vast integrated electric components. A sizable IRS's internal control is also challenging. In contrast to a single large IRS, several moderate-sized IRSs can lower production and control costs.

Drones offer two key benefits: a) variable positioning and 1142 b) dependable A2G communication linkages. As a result, 1143 drones may be deployed swiftly in disaster zones or hotspots 1144 to enable dependable wireless connectivity. In Figure 5, 1145 we consider a situation in which the combination of drone 1146 swarm and IRS is positioned to serve ground users in a 1147 circular area. Users may be present in hotspots or disaster 1148 areas, but impediments may prevent communication between 1149 the BS and users. Figure 5 shows how drone swarm -IRS 1150 can help in the disaster area for presentation and mitigate the 1151 impact of the disaster. 1152



FIGURE 5. Drone swarm meets IRS for large disaster area.

1153 F. APPLICATION OF DRONE S.WARM

¹¹⁵⁴ This section discusses the applications of drone swarm ¹¹⁵⁵ enabled IRS.

1156 1) MULTIPLE REFLECTIONS FOR REMOTE IOT

It is challenging for the present cellular network to reach 1157 outlying locations like forests, enormous oceans, volcanic 1158 lands, and other challenging conditions. IoT devices may be 1159 widely used for specialized activities requiring data transfers, 1160 such as integrating sensor data with high-definition sound 1161 and video information. Future wireless networks can use the 1162 combination of drone swarm and IRS to offer ubiquitous 1163 wireless connections and reliable data flow for distant IoT. 1164

Due to the significantly increased signal attenuation, pure 1165 multiples of the combination ofdrone swarm and IRS reflec-1166 tions are ineffective, according to the product distance-based 1167 path loss (i.e., twice path loss) model in IRS communication. 1168 Fortunately, drones with various characteristics may work 1169 together as a drone swarm to provide data transfers for distant 1170 IoT. As an illustration, the combination of drone and IRS can 1171 reflect the signal to a drone relay, which then decodes and 1172 transmits the signal to a second combination of drone and 1173 IRS. In this manner, BS and distant IoT devices are connected 1174 over a dependable wireless network as needed. 1175

2) A2A REFLECTION FOR DISASTER AREAS

Ground-based small cell BSs are often installed in densely 1177 populated regions like downtown, stadiums, and public 1178 spaces in cellular networks. Users in disaster areas can 1179 communicate wirelessly, in this case, thanks to the integration 1180 of IRS and drone swarm system. For instance, swarm 1181 drone-IRS reflects wireless signals from drone BS to users 1182 with a significant aperture gain. The signal strength under 1183 such A2A reflection is greatly increased compared to 1184 traditional BSs because of the dependable A2G pathways. 1185 A similar A2A reflection technology can also be used in 1186 disaster zones if the infrastructure on the ground is damaged. 1187

3) SIGNAL ENHANCEMENT IN DRONE SWARM

For information-sharing purposes, such as exchanging flight 1189 control signals, training an autonomous flight model, and 1190 offloading computing duties, the drone swarm may occasion-1191 ally need to connect to BS. Therefore, it is vital to guarantee 1192 a reliable wireless connection between the drone swarm and 1193 the BS. The integration of IRS and drone swarm can fly 1194 synchronously next to drone swarm users to deliver a reliable 1195 reflected signal for drone swarm users. The beamforming 1196 design for this application must consider the drones' velocity 1197 and movement. 1198

IV. FRAMEWORK FOR FL MEETS IRS IN DRONES TO SUPPORT 6G

The goal of FL meeting IRS in drones' framework to enable 1201 6G is to provide a novel solution for wireless communication 1202 networks. The framework for integrating FL and IRS in 1203 drones aims to increase the coverage area, enhance the quality 1204 of wireless communication services, provide reliable and 1205 efficient deployment of wireless communication services, 1206 and improve communication in dynamic environments. 1207 Briefly, the aim of the framework for FL meets IRS in 1208 drones for enabling 6G is to provide a highly optimized 1209 solution for wireless communication networks that offers 1210 improved coverage, quality, and reliability [100]. Due to 1211 the investigation of statistical training models directly on 1212 remote devices, FL has emerged as the focal point in the 1213 large-scale area of distributed optimization [101]. Figure 6 1214 illustrates a framework for combining FL and IRS in drones 1215 for supporting 6G. Combining IRS with drones offers a 1216 high potential for improving drone connection in various 1217

applications. Because drones have more flexibility and 1218 mobility, they can help optimize the IRS link by determining 1219 the best location for the IRS-drone. Nonetheless, the accuracy 1220 of the phase estimate and co-phasing operations is critical 1221 to getting the final benefit from the IRS. In practice, both 1222 activities are imperfect, so the IRS technology's eventual 1223 benefit may not be assured, which is especially important 1224 for IRS-drone setups. In this paper, we combine drone-IRSs 1225 and swarm drone-IRS to improve the performance of 6G 1226 networks. We highlight the desirable advantages of fusing 1227 IRS and drones. Then, we discuss the drone-IRS and swarm 1228 drone-IRS applications. In addition, the practical limits of 1229 the IRS and drone and the transmission architecture must be 1230 considered for efficient IRS-drone deployment. 1231

The FL no longer has a problem with personal data being 1232 accessible. In allocated networks, the FL approach is learned 1233 closer to devices. Because it protects privacy, the FL model 1234 may be used to create various IRS networks. IRS functions as 1235 a distributed trainer in this network to train the data generated 1236 and then create a model that transmits to an aggregating unit. 1237 In this fashion, decentralized FL learning for deployment and 1238 policy design is possible. Figure 7 illustrates an FL-based IRS 1239 networking system. Each thing in a smart environment has a local ML model that receives the learned parameter from the 1241 drone after being trained on a local dataset. The real-world 1242 applications that leverage the combination of FL and IRS in 1243 drone-enhanced communication networks are: 1244

Emergency response: In disaster-stricken areas, drones equipped with IRS can provide communication and sensing services, improving the coverage and reliability of the communication network. Edge devices, such as smartphones and laptops, can use FL to collaborate and learn a shared model, providing critical information and support to the rescue teams.

Agriculture: IRS can be used to enhance the connectivity and reliability of IoT devices in agriculture, such as sensors and drones, allowing them to communicate effectively and collect data from the fields. FL can be used to learn a shared model for IoT devices, providing insights and recommendations for improving crop yield and reducing waste.

Transportation: IRS can improve the connectivity and reliability of vehicles and communication infrastructure in transportation, such as traffic lights, road signs, and communication towers. FL can be used to learn a shared vehicle model, providing information and insights for improving the traffic flow, reducing accidents, and enhancing energy efficiency [102].

Industrial automation: IRS can be used to improve the connectivity and reliability of industrial automation systems, such as robots and sensors. FL can be used to learn a shared system model, providing information and insights for improving efficiency, reducing downtime, and enhancing safety. These are just a few examples of the potential applications of the combination of FL and IRS in drone-enhanced communication networks. The combination 1273 can benefit various domains, including communication, 1274 computation, sensing, and more, by providing a flexible and 1275 scalable solution for communication and computation. 1276

To boost the data rate in a single antenna with massive 1277 users, the authors [103] presented an IRS-assisted optimum 1278 beam reflection-FL. The experimental findings imply that the 1279 attainable rate is comparable to other centralized ML models 1280 and that changing the receiver number has no appreciable 1281 impact on the rate. To address resource allocation and 1282 device selection issues for aggregation accuracy and coverage 1283 rate improvements, the authors of [58] introduced an air 1284 FL system. The findings show that the suggested model 1285 can converge faster and with less training loss. A MIMO 1286 system was created using the suggested FL-convolutional 1287 neural network model. The received signals were utilized 1288 as an ML model to contribute to the application of drone 1289 trajectory [104]. 1290

A deep learning design method was developed to create the 1291 IRS configuration matrix, which used the sampled channel 1292 state information for the training of IRS [44]. However, 1293 preserving user privacy throughout the communication pro-1294 cess has not received enough attention in earlier works to 1295 be considered a critical concern. Therefore, FL has been 1296 suggested as a fresh approach to dealing with distributed 1297 learning's data privacy concerns in recent years. For instance, 1298 FL created low-latency Vehicle-to-Vehicle connections while 1299 safeguarding the users' private information [25]. The CSI 1300 between a user and the IRS, an IRS-assisted B5G system, 1301 is a class of private data closely tied to a user's location 1302 data. To accomplish high-speed communication with the CSI, 1303 optimum beam reflection based on FL is presented in [103]. 1304

FL is needed in IRS-enabled drone systems to enable local 1305 training of ML models without compromising data privacy 1306 and security. In drones, FL improves the performance and 1307 reliability of wireless communication by allowing each drone 1308 to train its model on local data, reducing the need to transmit 1309 sensitive data to a central server, which improves data privacy 1310 and reduces the risk of data breaches. FL features are helpful 1311 in IRS-enabled drones, including the ability to train ML 1312 models on local data, aggregate the model updates from 1313 multiple drones, and protect sensitive data through secure 1314 and privacy-preserving techniques. In addition, FL improves 1315 wireless communication's performance and reliability by 1316 enabling real-time decision-making and reducing the need for 1317 data transmission. 1318

In IRS-enabled drones using FL, sensitive data such 1319 as surveillance data or personal information needs to be 1320 protected to ensure privacy and prevent data breaches. 1321 Additionally, model parameters must be protected to avoid 1322 unauthorized access and ensure that the models are trained 1323 on accurate and representative data. The model information is 1324 exchanged through a central server on an IRS-enabled drone 1325 using FL. Each drone trains its local model on local data 1326 and sends the updated model parameters to the central server. 1327



FIGURE 6. Framework for FL meets IRS in drones for enabling 6G network.



FIGURE 7. Covergence of FL and IRS for efficient 6G network with drone help.

The central server aggregates the model updates and sends
 the updated global model back to the drones. The local data
 can include information about the wireless communication

environment, such as signal strength, interference, and other factors affecting communication performance. By using IRS to reflect and manipulate the signal, the proposed FL meets

IRS in drones, adapts to the changing environment, and 1334 optimizes wireless link quality, improving performance and 1335 reliability. The framework's effectiveness depends on several 1336 factors, including the system's complexity, the learning 1337 speed of FL, and the dynamic environment in which 1338 the drone system is used. The framework's effectiveness 1339 improves through careful design and implementation and the 1340 development of appropriate security and privacy techniques. 1341 Additionally, the learning speed of FL improves through 1342 advanced optimization algorithms and hardware acceleration 1343 techniques. 1344

FL for IRS deployment in UAV computing To address 1345 the increased capacity needs, future 6G will rely on 1346 high-frequency millimeter wave [105]. However, a funda-1347 mental design challenge for implementing mmWave bands 1348 into wide-scale commercial usage is permitting reliable 1349 mmWave connectivity under the obstruction. To solve the 1350 issue of blocking connections and boosting the power of the 1351 electromagnetic wave, passive reflectors have been proposed 1352 [106]. Specifically, using numerous reflectors increases the 1353 likelihood of LoS, lowering mmWave channel attenuation 1354 substantially. Several research [106], [107] have advocated 1355 using IRSs in mmWave. Still, these studies depend on the 1356 passive reflectors in a fixed and random position, which 1357 is unsatisfactory, provided the unpredictable changes of 1358 mmWave channels. 1359

Mobile reflectors, such as drone-carried IRS, are ideal 1360 for enhancing mmWave than stationary IRS because of 1361 the random nature of mmWave channels. The authors of 1362 [18] developed an IRS-enabled drone for establishing a 1363 LoS channel between BS and mobile users. In particular, 1364 a novel architecture for deploying a drone-enabled IRS to 1365 aid the transmission of mmWave downlink in a mobile and dynamic environment. Meanwhile, the scientists presented 1367 a framework for self-powering the IRS via radio-frequency 1368 energy harvesting. Simulation results indicated a consider-1369 able improvement in average data rate and attainable down-1370 link LoS probability when employing an IR-aided framework 1371 in a drone compared to a static IRS environment. In [18] 1372 introduced drone-IRS deployment for mmWave channels 1373 with radio-frequency energy harvesting using reinforcement 1374 learning. However, downlink transmission only examines a 1375 single user and ignores the more complex topic of multi-1376 user connections. In [108] simulated a multi-user IRS-carried 1377 drone, a distributional reinforcement learning approach was 1378 presented to optimize the reflection parameters, drone place-1379 ment, and precoding matrix at BS. In a multi-user setting, 1380 deep reinforcement learning techniques are used to discover 1381 the best deployment of a drone-IRS for effective downlink 1382 transmissions across mmWave frequencies. Compared to 1383 non-learning drone-IRS, IRS, and direct transmission, the 1384 findings demonstrated that deep reinforcement learning 1385 could learn the appropriate placement of the drone-IRS 1386 and achieve greater downlink capacity and an achievable 1387 rate. 1388

Unlike [18], Liu et al. [51] ignored energy harvesting and 1389 presented the challenge of reducing drone energy consump-1390 tion as a decaying deep O-network method. The NOMA 1391 for an IRS-empowered drone framework was implemented 1392 to improve user QoS. The challenge of minimizing energy 1393 consumption is defined as a combination of drone trajectory, 1394 power allocation strategy from drone to users and IRS phase 1395 shift. The energy dissipation of the drone may be significantly 1396 lowered by deploying drone-IRSs by adding NOMA and 1397 using 11.7% less energy than in the IRS-OMA scenario, 1398 according to numerical data. 1399

The authors of [82] investigated IoT traffic uplink trans-1400 mission in a drone-IRS system. To decrease the information 1401 average age, deep reinforcement learning based on protocol 1402 optimization was used to learn the unpredictability of IoT 1403 device activation patterns and manage the phase-shift, height 1404 of the drone, and communication scheduling of IRS. The 1405 authors established the drone's schedule and altitude in [82]. 1406 However, this study used only one drone, and trajectory 1407 optimization was not considered. The NOMA approach 1408 examined that it did not require a LoS channel between users 1409 and BS. Hariz et al., [109] investigated multiple drones' sub-1410 carrier distribution and trajectory to increase user coverage. 1411 NOMA examined the LoS link between the receiver and 1412 users, while NLoS was between the drone and users. The 1413 adopted double deep Q network approach is used to tackle 1414 the presented problem. The drone-IRS system may be used in 1415 IoT networks by adjusting power, sub-carrier, trajectory, and 1416 phase shift. Furthermore, the suggested technique reduces 1417 users' average information age while maintaining maximum 1418 transmit power and drone mobility limits. According to 1419 numerical data, the suggested technique outperforms the 1420 random-trajectory and matching algorithms by 15% and 1421 10%, respectively. Regarding IRS deployment in cutting-1422 edge networks, writers in [110] introduced high-speed trains 1423 and recommended a drone-IRS to offer high-speed trains 1424 robust and dependable communication services. The authors 1425 looked at the combined design of a phase shift and a drone 1426 trajectory and devised an actor-critic method to optimize 1427 high-speed trains' least feasible data rates. Compared to the 1428 IRS's random and fixed phase shift, the proposed method 1429 learns the best drone trajectory and IRS phase shift and 1430 achieves high data rates. 1431

FL meets IRS in drone swarm: FL and IRS in drone 1432 swarm enable 6G networks, a promising approach for 1433 improving wireless communication service coverage and 1434 quality. Multiple IRS-equipped drones are deployed in the 1435 communication environment to reflect incoming signals and 1436 form a wireless communication network. FL is used to 1437 continuously learn from the communication environment and 1438 optimize the reflection coefficients of each IRS. This allows 1439 the network to adapt to changing communication environ-1440 ments and improve the coverage and quality of wireless 1441 communication services. Furthermore, by integrating FL and 1442 IRS in drone swarm deployment, the framework enables 1443 the development of 6G networks, which provide higher data rates, lower latency, and more reliable communication services. As a result, this framework has the potential to transform wireless communication and provide new opportunities for communication-intensive applications and services, such as drone-based remote sensing, delivery, and inspection services.

Let W be the weight vector representing an IRS's reflection 1451 coefficients, and let L(W) be the loss function that measures 1452 the error between the desired signal and the received signal 1453 after reflection from the IRS. Federated learning aims to 1454 optimize the weight vector W across multiple devices (i.e., 1455 drones) without sharing the raw data. This is achieved through 1456 local model updates and global model aggregation. In each 1457 local model update, a device (i.e., a drone) computes the 1458 gradient of the loss function concerning its local data and 1459 sends it to the server. The server then aggregates the gradients 1460 from all devices and computes the global gradient. The 1461 weights of the IRS are updated using the computed global 1462 gradient as follows: 1463

$$W' = W - \eta \cdot \nabla L(W) \tag{6}$$

where is the learning rate that controls the step size of the updates and $\nabla L(W)$ is the global gradient of the loss function concerning the weight vector W. To ensure privacy, each device only sends the gradient of the loss function and not the raw data. The server updates the weights of the IRS without accessing the raw data. The algorithm flowchart process is given in the algorithm.2.

Algorithm 2	Processing	Flowchart
-------------	------------	-----------

1: Initialize the weight vector W of the IRS.

2: Repeat until convergence:

1464

a. Randomly sample a subset of devices.

b. For each device *i*, compute the gradient $L_i(W)$ of the loss function concerning its local data.

c. Send $L_i(W)$ to the server.

d. At the server, aggregate the gradients from all devices to compute the global gradient L(W).

e. Update the weight vector W using the computed global gradient: W' = W - *L(W)

3: Return the optimized weight vector W.

FL in IRS in swarm drones has two major goals: reduce 1472 signal distortion and increase FL convergence rate. In addi-1473 tion, IRS has been acknowledged as a revolutionary method 1474 to deftly change the complicated radio signal propagation 1475 environment by putting passively reflecting components in 1476 drone swarm on programmable surfaces [111]. In particular, 1477 even when IRSs in drone swarm are taken into account, 1478 IRSs can proactively adjust the wireless channels between 1479 the drone and smart devices by carefully managing each 1480 reflecting element's amplitude and phase shift in real-1481 time [112]. FL jointly improves model synchronization and 1482 the device employing IRSs in drone swarm to decrease 1483

propagation error while accelerating the convergence rate. 1484 IRSs are crucial in converting wireless channels into a usable 1485 computer to achieve FL's desired weighted sum feature. 1486 Furthermore, to effectively improve parameter aggregation 1487 from smart devices, drone swarm equipped with IRSs are 1488 used. Many unresolved concerns, such as the joint design of 1489 transmit reflect and receive in IRSs in drone swarm -assisted 1490 smart device networks, must still be resolved since FL in IRSs 1491 in drone swarm technology is still in its infancy. As illustrated 1492 in Figure 8, the FL satisfies IRS requirements for drones to 1493 gather data from smart devices and process that data locally 1494 in drones. The selection of devices to take part in the model 1495 uploading process, rather than averaging all local parameters 1496

FL and IRS in a drone swarm is a novel framework for 1497 enhancing the coverage and quality of wireless communi-1498 cation networks. In this framework, multiple IRS-equipped 1499 drones are deployed as a swarm in the communication 1500 environment to form a distributed wireless network. FL algo-1501 rithms coordinate the learning process among the drones 1502 and optimize the reflection coefficients of each IRS. This 1503 allows the swarm to adapt to changing communication 1504 environments and improve the coverage and quality of 1505 wireless communication services. Integrating FL and IRS 1506 technology in a drone swarm offers several advantages over 1507 traditional wireless communication networks. For example, 1508 the swarm rapidly deploys in disasters or emergencies, 1509 providing communication services to affected areas. The 1510 swarm offer communication services to remote or under-1511 served areas where traditional communication infrastructure 1512 is unavailable or unreliable. The framework of FL and 1513 IRS in a drone swarm has the potential to revolutionize 1514 wireless communication and enable new opportunities for 1515 communication-intensive applications and services. 1516

FL has emerged as a promising approach to enable 1517 ML on edge devices without sharing raw data. Combined 1518 with IRS, it can significantly enhance the performance and 1519 efficiency of 6G communication networks, especially in the 1520 context of drones. In this integration, FL can enable drones 1521 to learn collaboratively from their local data and improve 1522 their performance and efficiency in real-time. This can 1523 improve drones' communication performance and security, 1524 allowing for more efficient and secure data exchange. 1525 Additionally, IRS can reflect incoming signals to enhance 1526 the signal quality and reduce interference, further improving 1527 the communication performance of drones. The integration 1528 of FL and IRS in drones can bring several benefits for 6G 1529 communication networks, including: 1530

Improved communication performance: By enabling 1531 collaborative learning and intelligent signal reflection, integrating FL and IRS in drones can significantly improve the communication performance and reliability of 6G networks. 1534

Increased security: FL can help protect users' privacy by keeping the raw data on edge devices and only sharing the models. IRS can also enhance communication security by reducing the possibility of eavesdropping and interference.



FIGURE 8. FL in Multi-IRS in drones enabling 6G.

Reduced latency: By enabling real-time learning and 1540 adaptation, FL and IRS integration in drones can significantly 1541 reduce the latency of 6G communication networks, making 1542 them more suitable for latency-sensitive applications such as 1543 autonomous drones. 1544

Increased energy efficiency: By reducing the communi-1545 cation overhead and improving the signal quality, integrating 1546 FL and IRS in drones can significantly increase the energy 1547 efficiency of 6G communication networks, making them 1548 more sustainable and environmentally friendly. 1549

The integration of FL and IRS in drones has the potential 1550 to significantly enhance the performance and efficiency of 1551 6G communication networks. Furthermore, this integration 1552 can bring new opportunities and capabilities for various 1553 applications, such as autonomous drones, aerial photography 1554 and delivery services, environmental monitoring, disaster 1555 response, and infrastructure inspection. The algorithm flow 1556 charts for FL optimize the reflection coefficients is shown in 1557 algorithm 3. 1558

Mathematical expression for the loss function: The loss 1559 function can be defined as the sum of the mean squared error 1560 (MSE) between the received signal at each device and the 1561 desired signal, weighted by a regularization term: 1562

$$L = \sum_{i=1}^{N} w_i \cdot \text{MSE}(y_i, \hat{y}_i) + \lambda \cdot |\boldsymbol{w}|_2^2$$

where N is the number of devices, y_i is the desired signal at 1564 device *i*, \hat{y}_i is the received signal at device *i* after reflection 1565 from the IRS, w is the vector of IRS reflection coefficients, w_i 1566 is the weight assigned to device *i*, and λ is the regularization 1567 parameter. The loss function can be defined as the sum 1568 of the mean squared error (MSE) between the received 1569

1563

Algorithm 3 Algorithm Flow Chart for Optimizing the Reflection coefficients

1: Initialize the IRS reflection coefficients randomly.

2: Partition the devices into groups and send the IRS reflection coefficients to each group.

3: Each device uses its local data to compute the gradient of a loss function concerning the IRS reflection coefficients.

4: Each device sends the computed gradients to a central server.

5: The central server aggregates the gradients and computes the average gradient.

6: The central server updates the IRS reflection coefficients using the average gradient.

7: Repeat steps 2-6 until convergence.

signal at each device and the desired signal, weighted by a regularization term.

Mathematical expression for the gradient of the loss 1572 function: The gradient of the loss function concerning the 1573 IRS reflection coefficients can be computed as follows:

$$\nabla \mathcal{L} = \sum_{i=1}^{N} w_i \cdot \text{MSE}(\mathbf{y}_i, \hat{\mathbf{y}}_i) \cdot \mathbf{a}_i + 2\lambda \cdot \mathbf{w}$$
¹⁵⁷⁵

where a_i is the vector of the complex amplitudes of the signals 1576 the IRS reflects at device *i*. Including these details in the 1577 paper would provide a more detailed explanation of how FL 1578 optimizes the reflection coefficients of IRS in swarm drones. 1579

The proposed framework of FL meets IRS in drones for 1580 enabling 6G communication networks has several potential 1581 advantages and limitations. These can be summarized as 1582 follows: 1583

1570

1571

A. ADVANTAGES OF PROPOSED FRAMEWORK 1584

1) ENHANCED WIRELESS COMMUNICATION 1585

The integration of FL and IRS in drones can significantly 1586 enhance wireless communication in 6G networks. FL allows 1587 drones to collaboratively train ML models while keeping 1588 their data locally, preserving privacy and data ownership. IRS 1589 units strategically placed in the environment can manipulate 1590 the propagation of wireless signals, improving coverage, 1591 reliability, energy consumption, or costs. This combined 1592 approach can optimize wireless communication services 1593 and improve the overall performance of the communication 1594 network. 1595

2) DISTRIBUTED AND SCALABLE 1596

The proposed framework is distributed, with drones acting 1597 as data collectors, model trainers, and aggregators. This 1598 distributed approach allows for scalability, as more drones 1599 can be deployed in the network to increase coverage 1600 and capacity. FL enables drone collaboration without a 1601 centralized server, reducing communication overhead and 1602 enabling efficient communication in a large-scale network. 1603 This makes the framework suitable for dynamic and evolving 1604 environments like 6G communication networks. 1605

COST-EFFECTIVE 1606

The use of drones and IRS units in the framework has the 1607 potential to be cost-effective. Drones can be deployed flexibly 1608 and dynamically in the communication network, eliminating 1609 the need for fixed infrastructure. IRS units are passive and 1610 do not require active power consumption, making them 1611 energy-efficient and cost-effective compared to traditional 1612 communication infrastructure. This can result in cost savings 1613 in the communication network's deployment, operation, and 1614 maintenance. 1615

4) IMPROVED PRIVACY AND SECURITY 1616

FL allows drones to train ML models locally without sharing 1617 raw data, preserving privacy and data ownership. This can 1618 address privacy concerns associated with data sharing in 1619 wireless communication networks. Additionally, IRS units 1620 do not require data transmission or storage, reducing the 1621 risk of data breaches or cyber-attacks. This can result in 1622 improved privacy and security of communication within the 1623 framework. 1624

Summary: Integrating FL and IRS technology in drones 1625 offers a robust solution for enhancing wireless communi-1626 cation in 6G networks. The integration optimizes coverage, 1627 reliability, and energy efficiency by allowing drones to col-1628 laboratively train machine learning models while maintaining 1629 data privacy and leveraging IRS units to manipulate wireless 1630 signals strategically. The decentralized and scalable nature 1631 of the framework accommodates dynamic environments with 1632 the cost-effective deployment of drones and passive IRS 1633 units. Additionally, the combined approach enhances privacy 1634 and security by preserving data ownership, reducing data 1635

transmission risks, and presenting a compelling solution for 1636 advancing wireless communication systems. 1637

B. LIMITATIONS OF THE PROPOSED FRAMEWORK

1) COMPUTATIONAL AND ENERGY CONSTRAINTS OF DRONES

Drones may have limited computational and energy 1641 resources, affecting the performance of FL and IRS 1642 operations. Training ML models locally on drones can be 1643 computationally intensive and may require significant energy 1644 consumption, leading to reduced battery life and operational 1645 time. This can impact the scalability and performance of the 1646 framework. 1647

2) REGULATORY AND LEGAL CHALLENGES

Integrating drones with IRS in 6G communication networks 1649 may face regulatory and legal challenges, such as spec-1650 trum allocation, licensing, and compliance with aviation 1651 regulations. Regulatory frameworks for the operation of 1652 drones and IRS units may vary across different regions or 1653 countries, which can affect the deployment and operation of 1654 the proposed framework. 1655

3) PROPAGATION ENVIRONMENT LIMITATIONS

The performance of IRS units in manipulating the propa-1657 gation of wireless signals depends on environmental conditions, such as the placement of IRS units, obstacles, 1659 and interference. If the propagation environment is not 1660 conducive to IRS operations, the performance improvement 1661 in coverage, reliability, energy consumption, or costs may 1662 be limited. Coordination and Communication Overhead: The 1663 proposed framework may require coordination among drones 1664 and communication with the centralized server for model 1665 aggregation, which can introduce communication overhead 1666 and latency. Efficient coordination and communication 1667 among drones may be challenging in dynamic and changing 1668 environments, and communication delays or failures may 1669 impact the framework's performance. 1670

Summary: Implementing FL and IRS technology in 1671 drones encounters various challenges. Drones' limited 1672 computational and energy resources can hinder FL and 1673 IRS operations, potentially decreasing battery life and 1674 scalability due to intensive local model training. Regu-1675 latory and legal hurdles, including spectrum allocation, 1676 licensing, and compliance with aviation regulations, pose 1677 obstacles to integrating drones and IRS units in 6G net-1678 works, especially given regional variations. Coordinating 1679 drones and communication for model aggregation introduces 1680 overhead and latency, with communication challenges in 1681 dynamic environments potentially impacting the framework's 1682 efficiency.

V. CHALLENGES AND FUTURE TRENDS

This section outlines challenges and future trends of leverag-1685 ing IRS-drone for a 6G wireless network. 1686

1683 1684

1658

1656

1648

1638

1639

A. FUTURE DIRECTIONS 1687

ROBUSTNESS AND ADAPTABILITY 1688

Future research could explore techniques to make the 1689 proposed IRS-enabled drone system more robust and adapt-1690 able to changing communication environments, weather 1691 conditions, and mission requirements. This could involve 1692 developing algorithms or mechanisms that dynamically 1693 adjust the reflection coefficients of the IRS based on 1694 real-time feedback from the environment or incorporating 1695 ML techniques for improved adaptation and performance. 1696

2) SCALABILITY AND COMPLEXITY 1697

The scalability and complexity of the proposed framework 1698 could be further investigated. This could include exploring 1699 approaches to manage many drones and IRS units efficiently, 1700 optimizing the communication and coordination between 1701 drones and IRS units, and addressing challenges related to 1702 system complexity, computational overhead, and communi-1703 cation overhead. 1704

3) INTEROPERABILITY WITH OTHER GENERATIONS OF 1705 NETWORKS 1706

1707

The proposed IRS-enabled drone system could be integrated with networks of other generations, such as 5G or future 1708 6G networks. Future research could investigate techniques 1709 to enable seamless interoperability between the IRS-enabled 1710 drone system and other communication networks, such 1711 as cross-network resource management, network slicing, 1712 or inter-network coordination. 1713

ENERGY EFFICIENCY AND SUSTAINABILITY 1714

1715 Energy efficiency and sustainability are essential for drone systems. Future research could explore techniques to opti-1716 mize the energy consumption of the IRS-enabled drone 1717 system, such as energy-aware routing, power control, and 1718 energy harvesting. Additionally, investigating the system's 1719 environmental impact, such as carbon footprint and sustain-1720 ability, could be relevant in future research. 1721

5) SECURITY AND PRIVACY 1722

Security and privacy are critical aspects of any communi-1723 cation system. Future research could focus on developing 1724 robust security mechanisms to protect the IRS-enabled 1725 drone system against potential cyberattacks, unauthorized 1726 access, and privacy breaches. This could include encryption, 1727 authentication, and access control mechanisms tailored to the 1728 unique characteristics of the IRS-enabled drone system. 1729

6) REAL-WORLD IMPLEMENTATIONS AND FIELD TRIALS 1730

While the proposed framework may be based on theoretical 1731 or simulated evaluations, future research could focus on 1732 real-world implementations and field trials to validate the 1733 performance, feasibility, and practicality of the IRS-enabled 1734 drone system. It involves experimental setups, measurements, 1735

and evaluations in real-world scenarios to gain insights into 1736 the system's performance and potential limitations. 1737

1738

1755

1769

7) CHANNEL STATE INFORMATION

In particular, drone-IRS networks have variable channel cir-1739 cumstances and high mobility; channel estimation accuracy is 1740 essential for improving phase shifts and beamforming gain in 1741 IRS-aided communication networks. Additionally, increasing 1742 the number of IRSs deployed to increase the number of 1743 IRS for user links, phase shifts, drone-IRS channels, and 1744 predicted channel parameters. Due to the frequent pilot 1745 transmissions required for precise channel state information 1746 estimates, the challenges above might considerably lower 1747 system performance. Therefore, precise channel prediction 1748 becomes a crucial problem for practical communication due 1749 to the IRS's intrinsic passive character and lack of RF 1750 chains. To overcome the problems of applying advanced ML 1751 techniques like FL, deep neural network, and transfer learning 1752 to produce accurate channel state information with a lower 1753 overhead. 1754

8) THZ COMMUNICATIONS

To handle significant data rates, THz communications 1756 are expected to use the bandwidth in higher frequencies 1757 effectively. However, the number of RF chains will greatly 1758 expand in THz communication, leading to greater hardware 1759 and energy costs. Additionally, obstruction and propagation 1760 loss are higher on higher frequency channels like the THz 1761 channels. To address these complex problems effectively, IRS 1762 may be installed at advantageous sites, including BSs, drones, 1763 and mobile users, to establish a strong LoS. To accurately 1764 predict the channel state information, optimize beamforming 1765 signs and phase shift at IRS, and establish LoS to enhance 1766 SNR, AI techniques must be developed with the help of the 1767 digital twin concept. 1768

9) DRONE COMMUNICATION

In drone-assisted wireless systems, the IRS deployment 1770 strategy increases the design freedom of drone trajectories; 1771 however, as the actual channel gains between the drone 1772 and users rely on drone trajectory and precoding method. 1773 The precoding design of the multi-antenna configuration 1774 is closely related to the trajectory design of the drone. 1775 In actuality, developing an IRS' combined trajectory and pre-1776 coding design in a drone context presents several difficulties. 1777 First, the combined gains of channels from the drone to the 1778 users become spatially and frequency-selective due to the 1779 numerous reflected propagations provided by IRSs, which 1780 complicates the design of drone trajectory. Therefore, more 1781 study is still needed into deploying IRSs in complex and 1782 dynamic networks while maintaining appropriate fairness and 1783 accomplishing the sum-rate target of drones. Further research 1784 is required because precise channel tracking detection in THz 1785 communication makes compensating for the Doppler spread 1786 and delay more difficult. 1787

10) ENERGY CONSUMPTION 1788

Due to the lack of a power amplifier, an IRS needs 1789 an energy supply [4]. On the other hand, energy saving 1790 is also crucial because of the drone's inadequate battery 1791 endurance. Energy is frequently a significant barrier to drone 1792 flying length, performance, and battery life. The use of 1793 wireless charging for drones while in flight is one remedy. 1794 In addition, employing WPT methods can transmit the 1795 necessary energy for mission continuity using another drone. 1796 Researchers should thus create energy-efficient procedures 1797 and appropriate optimization frameworks to lower power 1798 consumption without compromising the effectiveness of 1799 IRS-assisted drone communication. 1800

11) OPTIMIZATION OF IRS-DRONE OPERATION WITH 1801 DIFFERENT WEATHER CONDITIONS 1802

It is crucial to optimise IRS-assisted drone communication 1803 when using drones in challenging situations like strong winds 1804 or rain. However, non-linear models make it challenging 1805 to optimize drone trajectory, IRS phase shift, and resource 1806 allocation in particular. Finding innovative design solutions 1807 with little complexity and effective performance is therefore 1808 advised. In this sense, tools for AI and ML are potential 1809 methods for effectively designing and optimizing these 1810 networks. These strategies are built on trustworthy, safe, 1811 powerful technologies to optimize difficult settings. Fur-1812 thermore, complex networks may be analyzed for improved 1813 secrecy performance using hybrid online and offline methods 1814 and data-driven models. However, other variables must be 1815 investigated, such as excessive energy usage, latency, and 1816 throughput. 1817

12) CSI 1818

For drone-IRS, drones have flexibility, high mobility features 1819 and unpredictable channel circumstances; accurate channel 1820 prediction is essential for maximizing the phase shifts and 1821 beamforming gain in IRS-aided communication networks. 1822 The number of IRSs deployed will increase the number 1823 of IRSs for user connections, phase shifts, drone-IRS 1824 channels, and predicted channel parameters. Due to the 1825 channel transmissions required for precise CSI estimates, 1826 the challenges above might considerably lower network 1827 performance. Therefore, due to the IRS's intrinsic passive 1828 character and lack of RF chains, a precise channel estimate 1829 becomes a crucial problem for practical communication. 1830 To overcome challenges, it is necessary to use cutting-edge 1831 ML techniques such as FL, transfer learning, and DNNs to 1832 acquire accuracy. 1833

13) DATA GATHERING AND TRAINING MODEL 1834

Data gathering is an essential step in training ML, and the 1835 model quality depends on the quality of data [113]. However, 1836 data gathering is a barrier to applying ML-based approaches 1837 to IRS-based communication since partial data might lead to 1838 poor models. The key estimating elements are signal detec-1839

tion, channel estimation, and the receiver's beamforming 1840 architecture. Therefore, data-gathering methods may be a 1841 future study area for the practical application of ML-based 1842 strategies. 1843

14) SECURING DATA COLLECTION

The IRS system should be mobile enough to be mounted on 1845 a drone, as was recently investigated [52], to be placed at 1846 the ideal location. One significant disadvantage of drone-IRS 1847 is that hostile users can create an LoS link using the 1848 non-specific nature of the reflected radio signal, compro-1849 mising the communication's confidentiality. The influence of 1850 a drone-IRS system on secure data transmission rates from 1851 smart environments was the main emphasis in [114], which 1852 sought to maximize the feasible secrecy rates under total 1853 transmit power constraints. 1854

15) PERFORMANCE OPTIMIZATION

Further research can be conducted to optimize the per-1856 formance of IRS-enabled drone systems regarding signal quality, coverage, capacity, and energy efficiency. This can 1858 involve exploring novel algorithms, techniques, and architec-1859 tures for jointly optimizing the operation of drones and IRS 1860 in dynamic and changing communication environments.

16) INTEROPERABILITY WITH OTHER NETWORKS

Investigation can be done on how IRS-enabled drone systems 1863 can interoperate with other networks of different generations, 1864 such as 5G and beyond, to enable seamless communication 1865 and networking across heterogeneous networks. This can 1866 involve exploring interoperability protocols, handover mech-1867 anisms, and network management strategies to ensure smooth 1868 integration and operation with other networks. 1869

17) ROBUSTNESS AND RESILIENCE

Research can be conducted to enhance the robustness and 1871 resilience of IRS-enabled drone systems against various 1872 challenges, such as interference, jamming, mobility, and envi-1873 ronmental conditions. This can involve investigating adaptive 1874 algorithms, distributed coordination, and fault-tolerant mech-1875 anisms to ensure the reliable and resilient operation of the 1876 system in dynamic and hostile environments. 1877

18) SECURITY AND PRIVACY

Further investigation can be done on the security and 1879 privacy aspects of IRS-enabled drone systems, including 1880 protecting against unauthorized access, data breaches, and 1881 privacy violations. This can involve exploring encryption, 1882 authentication, and access control mechanisms tailored for 1883 IRS-enabled drone systems to ensure secure and privacy-1884 preserving communication. 1885

19) REGULATION AND STANDARDIZATION

Research can be conducted on the regulatory and standard-1887 ization aspects of IRS-enabled drone systems, including 1888

1857

1855

1844

1861

1862

1870

1878

addressing legal, ethical, and policy issues related to their 1889 deployment, operation, and management. This can involve 1890 studying regulatory frameworks, policy guidelines, and stan-1891 dardization efforts to ensure compliance and harmonization 1892 with relevant regulations and standards. These are just generic 1893 suggestions, and the specific areas of improvement and future 1894 research would depend on the findings, limitations, and 1895 contributions of the specific paper you mentioned, as well 1896 as the research objectives and context of the proposed 1897 framework for IRS-enabled drone systems. It is important 1898 for the paper's authors to carefully consider their specific 1899 research findings and contributions and provide relevant and 1900 meaningful suggestions for future research based on their 1901 work 1902

1903 **B. CHALLANGES**

The potential challenges and open issues related to integrating
 IRS with drones in 6G communication are summaries fellow.

1906 1) TECHNICAL CHALLENGES

Integrating IRS with drones in 6G communication may pose 1907 various technical challenges. Advanced signal processing 1908 algorithms may be required to optimize the reflections 1909 from the IRS units to achieve the desired communication 1910 performance. Efficient communication protocols need to 1911 be designed to enable effective coordination and informa-1912 tion exchange between drones and IRS units. Accurate 1913 localization and tracking techniques are crucial for the 1914 precise positioning and movement of drones and IRS units, 1915 especially in dynamic and changing environments. The 1916 authors could discuss the technical challenges associated 1917 with these aspects, including developing novel algorithms, 1918 protocols, and techniques to address them and the potential 1919 impact on the overall system performance. 1920

1921 2) REGULATORY CHALLENGES

Integrating IRS with drones in 6G communication may 1922 also face regulatory challenges. Spectrum allocation for 1923 communication between drones and IRS units may need 1924 to be carefully considered, considering the availability, 1925 compatibility, and interference issues related to the spectrum 1926 bands used. Licensing requirements for operating drones 1927 and IRS units, compliance with aviation regulations, and 1928 other regulatory considerations may impact the deployment 1929 and operation of the system. The authors could discuss 1930 the regulatory challenges and requirements associated with 1931 integrating the IRS with drones and potential solutions or 1932 recommendations to address them. 1933

1934 3) PRIVACY CONCERNS AND SECURITY ISSUES

Privacy and security issues may arise in integrating IRS
with drones in 6G communication. The reflections from
the IRS units could reveal sensitive information about the
environment, infrastructure, or users. Secure communication,
authentication, and data privacy mechanisms may be required
to protect the communications' integrity, confidentiality,

and privacy between drones and IRS units. The authors 1941 could discuss the potential privacy concerns and security 1942 issues associated with the system and propose appropriate 1943 measures or techniques to mitigate them, such as encryption, 1944 authentication, and access control mechanisms. 1945

1946

1050

1960

1981

4) INTEROPERABILITY CHALLENGES

Interoperability challenges may arise in integrating IRS with 1947 existing networks or coexisting with other wireless tech-1948 nologies. Integration with existing communication networks, 1949 such as 5G or legacy networks, may require interoperability 1950 mechanisms and protocols to enable seamless communica-1951 tion between drones, IRS units, and other network entities. 1952 Coexistence with other wireless technologies or devices, 1953 such as Wi-Fi, cellular networks, or other drones, may pose 1954 interference or coordination challenges. The authors could 1955 discuss the interoperability challenges and potential solutions 1956 to ensure smooth integration and coexistence with other 1957 communication technologies or networks. 1958

5) PRIVACY AND SECURITY IN FL FOR UAV-ENABLED NETWORKS

Ensuring privacy and security in FL for UAV-enabled net-1961 works presents multifaceted challenges. FL's decentralized 1962 approach, where UAVs train models locally without sharing 1963 raw data, requires addressing data privacy, model confi-1964 dentiality, and secure communication issues. Safeguarding 1965 against potential data breaches during model aggregation, 1966 protecting against adversarial attacks, and preventing model 1967 poisoning while maintaining the integrity of the learning 1968 process are crucial concerns. Additionally, achieving differ-1969 ential privacy across diverse data distributions and complying 1970 with regulatory frameworks further complicate the security 1971 landscape. Balancing the benefits of FL's decentralized model 1972 with robust privacy and security measures is essential for 1973 harnessing its potential in UAV-enabled networks while 1974 mitigating risks associated with data leakage, adversarial 1975 manipulation, and regulatory non-compliance. Leveraging 1976 FL to facilitate privacy-preserving collaboration among 1977 UAVs is for efficient learning, scheduling, and resource 1978 management in dynamic and privacy-sensitive environments 1979 [115], [116], [117] 1980

VI. CONCLUSION

The paper has addressed establishing a non-terrestrial 1982 network for 6G communications by presenting cutting-edge 1983 advancements in IRS and drone communication technolo-1984 gies. A key innovation lies in integrating IRS and FL 1985 within drones, offering a compelling avenue for enhancing 1986 6G communication network performance. The proposed 1987 framework showcases how the fusion can ameliorate wire-1988 less communication services, elevating coverage, reliability, 1989 and energy efficiency. The framework fosters collaborative 1990 learning among multiple drones, culminating in superior 1991 and more streamlined decision-making processes within 1992 the network. Amid the evident benefits, challenges like 1993

2061

2062

2063

2064

2065

2066

2067

2068

2069

2070

2071

2072

2073

2074

2075

2076

2077

2078

2079

2080

2081

2082

2083

2084

2085

2086

2087

2088

2089

2090

2091

2092

2093

2094

2095

2096

2097

2098

2099

2100

2101

2102

2103

2104

2105

2106

2107

2108

2109

2110

2111

2112

2113

2114

2118

2123

2124

2125

2126

2127

2128

2129

2130

2131

2132

2133

2134

regulatory and security aspects need to be resolved to harness 1994 this technology's full potential. Despite the challenges, the 1995 convergence of FL and IRS within drones holds substantial 1996 promise in catalyzing innovation for the evolution of 6G com-1997 munication networks and fulfilling the evolving requirements 1998 of future wireless communication services. A comprehensive 1999 exploration of various FL optimization techniques and 2000 algorithms proposed for UAV-enabled networks could offer 2001 valuable insights into the technical strategies driving the 2002 synergy of FL and IRS, enriching the overall framework's 2003 robustness and performance optimization. 2004

REFERENCES 2005

2016

2017

2018

2019

2020

2021

2022

2023

- [1] M. Mozaffari, W. Saad, M. Bennis, Y.-H. Nam, and M. Debbah, "A tutorial 2006 2007 on UAVs for wireless networks: Applications, challenges, and open problems," IEEE Commun. Surveys Tuts., vol. 21, no. 3, pp. 2334-2360, 2008 3rd Quart., 2019. 2009
- 2010 [2] A. A. Al-Habob, O. A. Dobre, S. Muhaidat, and H. V. Poor, "Energyefficient data dissemination using a UAV: An ant colony approach," IEEE 2011 Wireless Commun. Lett., vol. 10, no. 1, pp. 16-20, Jan. 2021. 2012
- [3] Y. Zeng, Q. Wu, and R. Zhang, "Accessing from the sky: A tutorial on 2013 UAV communications for 5G and beyond," Proc. IEEE, vol. 107, no. 12, 2014 pp. 2327-2375, Dec. 2019. 2015
 - [4] X. Pang, M. Sheng, N. Zhao, J. Tang, D. Niyato, and K.-K. Wong, "When UAV meets IRS: Expanding air-ground networks via passive reflection," IEEE Wireless Commun., vol. 28, no. 5, pp. 164-170, Oct. 2021.
 - [5] M. Zeng, X. Li, G. Li, W. Hao, and O. A. Dobre, "Sum rate maximization for IRS-assisted uplink NOMA," IEEE Commun. Lett., vol. 25, no. 1, pp. 234-238, Jan. 2021.
 - Q. Wu and R. Zhang, "Towards smart and reconfigurable environment: [6] Intelligent reflecting surface aided wireless network," IEEE Commun. Mag., vol. 58, no. 1, pp. 106-112, Jan. 2020.
- [7] S. H. Alsamhi, F. Afghah, R. Sahal, A. Hawbani, A. A. Al-qaness, B. Lee, 2025 2026 and M. Guizani, "Green IoT using UAVs in B5G networks: A review of applications and strategies," 2021, arXiv:2103.17043. 2027
- A. Saif, K. Dimyati, K. A. Noordin, N. S. M. Shah, S. H. Alsamhi, [8] 2028 and Q. Abdullah, "Energy-efficient tethered UAV deployment in B5G for 2029 2030 smart environments and disaster recovery," in Proc. 1st Int. Conf. Emerg. Smart Technol. Appl. (eSmarTA), Aug. 2021, pp. 1-5. 2031
- F. A. Almalki, B. O. Soufiene, S. H. Alsamhi, and H. Sakli, "A low-cost 2032 [9] 2033 platform for environmental smart farming monitoring system based on IoT and UAVs," Sustainability, vol. 13, no. 11, p. 5908, May 2021. 2034
- [10] S. H. Alsamhi, F. Afghah, R. Sahal, A. Hawbani, M. A. A. Al-Qaness, 2035 B. Lee, and M. Guizani, "Green Internet of Things using UAVs in B5G 2036 networks: A review of applications and strategies," Ad Hoc Netw., vol. 117, 2037 2038 Jun. 2021, Art. no. 102505.
- [11] S. Basharat, S. A. Hassan, H. Pervaiz, A. Mahmood, Z. Ding, and 2039 2040 M. Gidlund, "Reconfigurable intelligent surfaces: Potentials, applications, and challenges for 6G wireless networks," IEEE Wireless Commun., 2041 2042 vol. 28, no. 6, pp. 184-191, Dec. 2021.
- A. C. Pogaku, D.-T. Do, B. M. Lee, and N. D. Nguyen, "UAV-assisted 2043 [12] RIS for future wireless communications: A survey on optimization and 2044 performance analysis," IEEE Access, vol. 10, pp. 16320-16336, 2022. 2045
- [13] D. Ma, M. Ding, and M. Hassan, "Enhancing cellular communications for 2046 UAVs via intelligent reflective surface," in Proc. IEEE Wireless Commun. 2047 Netw. Conf. (WCNC), May 2020, pp. 1-6. 2048
- 2049 [14] M. Al-Jarrah, A. Al-Dweik, E. Alsusa, Y. Iraqi, and M.-S. Alouini, "On the performance of IRS-assisted multi-layer UAV communications 2050 2051 with imperfect phase compensation," IEEE Trans. Commun., vol. 69, no. 12, pp. 8551-8568, Dec. 2021. 2052
- [15] X. Pang, N. Zhao, J. Tang, C. Wu, D. Niyato, and K.-K. Wong, "IRS-2053 assisted secure UAV transmission via joint trajectory and beamforming 2054 design," IEEE Trans. Commun., vol. 70, no. 2, pp. 1140-1152, Feb. 2022. 2055
- S. Li, B. Duo, X. Yuan, Y.-C. Liang, and M. Di Renzo, "Reconfigurable 2056 [16] intelligent surface assisted UAV communication: Joint trajectory design 2057 and passive beamforming," IEEE Wireless Commun. Lett., vol. 9, no. 5, 2058 pp. 716-720, May 2020. 2059

- [17] S. Fang, G. Chen, and Y. Li, "Joint optimization for secure intelligent reflecting surface assisted UAV networks," IEEE Wireless Commun. Lett., vol. 10, no. 2, pp. 276-280, Feb. 2021.
- [18] Q. Zhang, W. Saad, and M. Bennis, "Reflections in the sky: Millimeter wave communication with UAV-carried intelligent reflectors," in Proc. IEEE Global Commun. Conf. (GLOBECOM), Dec. 2019, pp. 1-6.
- [19] S. Jiao, F. Fang, X. Zhou, and H. Zhang, "Joint beamforming and phase shift design in downlink UAV networks with IRS-assisted NOMA," J. Commun. Inf. Netw., vol. 5, no. 2, pp. 138-149, Jun. 2020.
- [20] A. S. Abdalla, K. Powell, V. Marojevic, and G. Geraci, "UAV-assisted attack prevention, detection, and recovery of 5G networks," IEEE Wireless Commun., vol. 27, no. 4, pp. 40-47, Aug. 2020.
- [21] Z. Mohamed and S. Aïssa, "Leveraging UAVs with intelligent reflecting surfaces for energy-efficient communications with cell-edge users," in Proc. IEEE Int. Conf. Commun. Workshops (ICC Workshops), Jun. 2020, pp. 1-6.
- [22] M. Hua, L. Yang, Q. Wu, C. Pan, C. Li, and A. L. Swindlehurst, "UAVassisted intelligent reflecting surface symbiotic radio system," IEEE Trans. Wireless Commun., vol. 20, no. 9, pp. 5769-5785, Sep. 2021.
- [23] L. Ge, P. Dong, H. Zhang, J.-B. Wang, and X. You, "Joint beamforming and trajectory optimization for intelligent reflecting surfaces-assisted UAV communications," IEEE Access, vol. 8, pp. 78702-78712, 2020.
- [24] T. Li, A. K. Sahu, A. Talwalkar, and V. Smith, "Federated learning: Challenges, methods, and future directions," IEEE Signal Process. Mag., vol. 37, no. 3, pp. 50-60, May 2020.
- S. Samarakoon, M. Bennis, W. Saad, and M. Debbah, "Federated learning [25] for ultra-reliable low-latency V2V communications," in Proc. IEEE Global Commun. Conf. (GLOBECOM), Dec. 2018, pp. 1-7.
- [26] T. Zhang and S. Mao, "Energy-efficient federated learning with intelligent reflecting surface," IEEE Trans. Green Commun. Netw., vol. 6, no. 2, pp. 845-858, Jun. 2022.
- [27] S. H. Alsamhi, O. Ma, M. S. Ansari, and F. A. Almalki, "Survey on collaborative smart drones and Internet of Things for improving smartness of smart cities," IEEE Access, vol. 7, pp. 128125-128152, 2019.
- [28] A. Gupta, S. Sundhan, S. K. Gupta, S. H. Alsamhi, and M. Rashid, "Collaboration of UAV & HetNet for better QoS: A comparative study," Int. J. Vehicle Inf. Commun. Syst., vol. 5, no. 3, pp. 309-333, Nov. 2020.
- [29] S. Alsamhi, O. Ma, M. Ansari, and S. Gupta, "Collaboration of drone and Internet of Public Safety Things in smart cities: An overview of QoS and network performance optimization," Drones, vol. 3, no. 1, p. 13, Jan. 2019.
- [30] S. Gopi, M. Magarini, S. Alsamhi, and A. Shvetsov, "Machine learningassisted adaptive modulation for optimized drone-user communication in B5G," Drones, vol. 5, no. 4, p. 128, Oct. 2021.
- [31] S. H. Alsamhi, B. Lee, M. Guizani, N. Kumar, Y. Qiao, and X. Liu, "Blockchain for decentralized multi-drone to combat COVID-19 and future pandemics: Framework and proposed solutions," Trans. Emerg. Telecommun. Technol., vol. 32, no. 9, Sep. 2021, Art. no. e4255.
- [32] A. Saif, K. Dimyati, K. A. Noordin, S. H. Alsamhi, and A. Hawbani, "Multi-UAV and SAR collaboration model for disaster management in B5G networks," Internet Technol. Lett., Jul. 2021, Art. no. e310, doi: 10.1002/itl2.310
- [33] S. H. Alsamhi, F. A. Almalki, O. Ma, M. S. Ansari, and B. Lee, "Predictive estimation of optimal signal strength from drones over IoT frameworks in smart cities," IEEE Trans. Mobile Comput., vol. 22, no. 1, pp. 402-416, Jan. 2023.
- [34] A. Saif, K. Dimvati, K. A. Noordin, S. H. Alsamhi, N. A. Mosali, and 2115 S. K. Gupta, "UAV and relay cooperation based on RSS for extending 2116 smart environments coverage area in B5G," 2022, doi: 10.21203/ 2117 rs.3.rs-2002265/v1.
- [35] S. H. Alsamhi, F. A. Almalki, F. Afghah, A. Hawbani, A. V. Shvetsov, 2119 B. Lee, and H. Song, "Drones' edge intelligence over smart environments 2120 in B5G: Blockchain and federated learning synergy," IEEE Trans. Green 2121 Commun. Netw., vol. 6, no. 1, pp. 295-312, Mar. 2022. 2122
- [36] S. H. Alsamhi, A. V. Shvetsov, S. Kumar, S. V. Shvetsova, M. A. Alhartomi, A. Hawbani, N. S. Rajput, S. Srivastava, A. Saif, and V. O. Nyangaresi, "UAV computing-assisted search and rescue mission framework for disaster and harsh environment mitigation," Drones, vol. 6, no. 7, p. 154, Jun. 2022.
- [37] S. Sundhan, A. Gupta, S. H. Alsamhi, and S. K. Gupta, "Survey on significance of UAV assisted random heterogeneous network," in Proc. Springer Int. Conf. Opt. Wireless Technol. (OWT), 2018, pp. 55-61.
- S. H. Alsamhi, A. V. Shvetsov, S. Kumar, J. Hassan, M. A. Alhartomi, [38] S. V. Shvetsova, R. Sahal, and A. Hawbani, "Computing in the sky: A survey on intelligent ubiquitous computing for UAV-assisted 6G networks and Industry 4.0/5.0," Drones, vol. 6, no. 7, p. 177, Jul. 2022.

- [39] A. Saif, K. Dimyati, K. A. Noordin, N. S. Mohd Shah, S. H. Alsamhi,
 Q. Abdullah, and N. Farah, "Distributed clustering for user devices under
 UAV coverage area during disaster recovery," in *Proc. IEEE Int. Conf. Power Eng. Appl. (ICPEA)*, Mar. 2021, pp. 143–148.
- [40] S. H. Alsamhi, O. Ma, and M. S. Ansari, "Predictive estimation of the optimal signal strength from unmanned aerial vehicle over Internet of Things using ANN," 2018, *arXiv:1805.07614*.
- [41] S. H. Alsamhi, F. A. Almalki, H. AL-Dois, A. V. Shvetsov, M. S. Ansari,
 A. Hawbani, S. K. Gupta, and B. Lee, "Multi-drone edge intelligence and
 SAR smart wearable devices for emergency communication," *Wireless Commun. Mobile Comput.*, vol. 2021, Nov. 2021, Art. no. 6710074.
- [42] A. Saif, K. Dimyati, K. A. Noordin, N. A. Mosali, D. Gc, and
 S. H. Alsamhi, "Skyward bound: Empowering disaster resilience with multi-UAV-assisted B5G networks for enhanced connectivity and energy efficiency," *Internet Things*, vol. 23, Oct. 2023, Art. no. 100885.
- [43] S. H. Alsamhi, A. V. Shvetsov, S. V. Shvetsova, A. Hawbani, M. Guizani,
 M. A. Alhartomi, and O. Ma, "Blockchain-empowered security and energy efficiency of drone swarm consensus for environment exploration," *IEEE Trans. Green Commun. Netw.*, vol. 7, no. 1, pp. 328–338,
 Mar. 2023.
- [44] M. Cui, G. Zhang, and R. Zhang, "Secure wireless communication via intelligent reflecting surface," *IEEE Wireless Commun. Lett.*, vol. 8, no. 5, pp. 1410–1414, Oct. 2019.
- [45] H. Shen, W. Xu, S. Gong, Z. He, and C. Zhao, "Secrecy rate maximization for intelligent reflecting surface assisted multi-antenna communications," *IEEE Commun. Lett.*, vol. 23, no. 9, pp. 1488–1492, Sep. 2019.
- [46] J. Chen, Y.-C. Liang, Y. Pei, and H. Guo, "Intelligent reflecting surface:
 A programmable wireless environment for physical layer security," *IEEE Access*, vol. 7, pp. 82599–82612, 2019.
- [47] X. Guan, Q. Wu, and R. Zhang, "Intelligent reflecting surface assisted secrecy communication: Is artificial noise helpful or not?" *IEEE Wireless Commun. Lett.*, vol. 9, no. 6, pp. 778–782, Jun. 2020.
- [48] H. Yang, Z. Xiong, J. Zhao, D. Niyato, L. Xiao, and Q. Wu, "Deep reinforcement learning-based intelligent reflecting surface for secure wireless communications," *IEEE Trans. Wireless Commun.*, vol. 20, no. 1, pp. 375–388, Jan. 2021.
- [49] W. Feng, J. Tang, Y. Yu, J. Song, N. Zhao, G. Chen, K.-K. Wong, and J. Chambers, "UAV-enabled SWIPT in IoT networks for emergency communications," *IEEE Wireless Commun.*, vol. 27, no. 5, pp. 140–147, Oct. 2020.
- [50] T. Shafique, H. Tabassum, and E. Hossain, "Optimization of wireless relaying with flexible UAV-borne reflecting surfaces," *IEEE Trans. Commun.*, vol. 69, no. 1, pp. 309–325, Jan. 2021.
- X. Liu, Y. Liu, and Y. Chen, "Machine learning empowered trajectory and passive beamforming design in UAV-RIS wireless networks," *IEEE J. Sel. Areas Commun.*, vol. 39, no. 7, pp. 2042–2055, Jul. 2021.
- [52] H. Lu, Y. Zeng, S. Jin, and R. Zhang, "Aerial intelligent reflecting surface: Joint placement and passive beamforming design with 3D beam flattening," *IEEE Trans. Wireless Commun.*, vol. 20, no. 7, pp. 4128–4143, Jul. 2021.
- [53] H. Long, M. Chen, Z. Yang, B. Wang, Z. Li, X. Yun, and M. Shikh-Bahaei,
 "Reflections in the sky: Joint trajectory and passive beamforming design
 for secure UAV networks with reconfigurable intelligent surface," 2020,
 arXiv:2005.10559.
- [54] W. Tang, M. Z. Chen, X. Chen, J. Y. Dai, Y. Han, M. Di Renzo, Y. Zeng, S. Jin, Q. Cheng, and T. J. Cui, "Wireless communications with reconfigurable intelligent surface: Path loss modeling and experimental measurement," *IEEE Trans. Wireless Commun.*, vol. 20, no. 1, pp. 421–439, Jan. 2021.
- [55] Z. Kang, C. You, and R. Zhang, "3D placement for multi-UAV relaying:
 An iterative Gibbs-sampling and block coordinate descent optimization approach," *IEEE Trans. Commun.*, vol. 69, no. 3, pp. 2047–2062, Mar. 2021.
- [56] I. Valiulahi and C. Masouros, "Multi-UAV deployment for throughput maximization in the presence of co-channel interference," *IEEE Internet Things J.*, vol. 8, no. 5, pp. 3605–3618, Mar. 2021.
- [57] M. Chen, Z. Yang, W. Saad, C. Yin, H. V. Poor, and S. Cui, "A joint learning and communications framework for federated learning over wireless networks," *IEEE Trans. Wireless Commun.*, vol. 20, no. 1, pp. 269–283, Jan. 2021.
- [58] W. Ni, Y. Liu, Z. Yang, H. Tian, and X. Shen, "Federated learning in multi-RIS-aided systems," *IEEE Internet Things J.*, vol. 9, no. 12, pp. 9608–9624, Jun. 2022.

- [59] S. H. Alsamhi, A. V. Shvetsov, A. Hawbani, S. V. Shvetsova, S. Kumar, and L. Zhao, "Survey on federated learning enabling indoor navigation for Industry 4.0 in B5G," *Future Gener. Comput. Syst.*, vol. 148, pp. 250–265, Nov. 2023.
- [60] F. C. Okogbaa, Q. Z. Ahmed, F. A. Khan, W. B. Abbas, F. Che, S. A. R. Zaidi, and T. Alade, "Design and application of Intelligent Reflecting Surface (IRS) for beyond 5G wireless networks: A review," *Sensors*, vol. 22, no. 7, p. 2436, Mar. 2022.
- [61] Z. Ding, L. Lv, F. Fang, O. A. Dobre, G. K. Karagiannidis, N. Al-Dhahir, R. Schober, and H. V. Poor, "A state-of-the-art survey on reconfigurable intelligent surface-assisted non-orthogonal multiple access networks," *Proc. IEEE*, vol. 110, no. 9, pp. 1358–1379, Sep. 2022.
- [62] B. Brik, A. Ksentini, and M. Bouaziz, "Federated learning for UAVsenabled wireless networks: Use cases, challenges, and open problems," *IEEE Access*, vol. 8, pp. 53841–53849, 2020.
- [63] M.-A. Lahmeri, M. A. Kishk, and M.-S. Alouini, "Artificial intelligence for UAV-enabled wireless networks: A survey," *IEEE Open J. Commun. Soc.*, vol. 2, pp. 1015–1040, 2021.
- [64] C. You, Z. Kang, Y. Zeng, and R. Zhang, "Enabling smart reflection in integrated air-ground wireless network: IRS meets UAV," *IEEE Wireless Commun.*, vol. 28, no. 6, pp. 138–144, Dec. 2021.
- [65] A. Taneja, S. Rani, and N. Herencsar, "Energy aware solution for IRSaided UAV communication in 6G wireless networks," *Sustain. Energy Technol. Assessments*, vol. 58, Aug. 2023, Art. no. 103318.
- [66] M. Ahmed, H. M. Alshahrani, N. Alruwais, M. M. Asiri, M. A. Duhayyim, W. U. Khan, T. Khurshaid, and A. Nauman, "Joint optimization of UAV-IRS placement and resource allocation for wireless powered mobile edge computing networks," *J. King Saud Univ. Comput. Inf. Sci.*, vol. 35, no. 8, Sep. 2023, Art. no. 101646.
- [67] B. G. Sheena and N. Snehalatha, "Multi-objective metaheuristic optimization-based clustering with network slicing technique for Internet of Things-enabled wireless sensor networks in 5G systems," *Trans. Emerg. Telecommun. Technol.*, vol. 34, no. 8, Aug. 2023, Art. no. e4626.
- [68] Z. Zhao, C. Feng, H. H. Yang, and X. Luo, "Federated-learning-enabled intelligent fog radio access networks: Fundamental theory, key techniques, and future trends," *IEEE Wireless Commun.*, vol. 27, no. 2, pp. 22–28, Apr. 2020.
- [69] L. Zhang, X. Q. Chen, S. Liu, Q. Zhang, J. Zhao, J. Y. Dai, G. D. Bai, X. Wan, Q. Cheng, G. Castaldi, V. Galdi, and T. J. Cui, "Space-time-coding digital metasurfaces," *Nature Commun.*, vol. 9, no. 1, pp. 1–11, Oct. 2018.
- [70] A. Díaz-Rubio, V. S. Asadchy, A. Elsakka, and S. A. Tretyakov, "From the generalized reflection law to the realization of perfect anomalous reflectors," *Sci. Adv.*, vol. 3, no. 8, Aug. 2017, Art. no. e1602714.
- [71] H. Guo, Y.-C. Liang, J. Chen, and E. G. Larsson, "Weighted sumrate maximization for intelligent reflecting surface enhanced wireless networks," in *Proc. IEEE Global Commun. Conf. (GLOBECOM)*, Dec. 2019, pp. 1–6.
- [72] C. Huang, S. Hu, G. C. Alexandropoulos, A. Zappone, C. Yuen, R. Zhang, M. D. Renzo, and M. Debbah, "Holographic MIMO surfaces for 6G wireless networks: Opportunities, challenges, and trends," *IEEE Wireless Commun.*, vol. 27, no. 5, pp. 118–125, Oct. 2020.
- [73] X. Tan, Z. Sun, J. M. Jornet, and D. Pados, "Increasing indoor spectrum sharing capacity using smart reflect-array," in *Proc. IEEE Int. Conf. Commun. (ICC)*, May 2016, pp. 1–6.
- [74] Y. Tang, G. Ma, H. Xie, J. Xu, and X. Han, "Joint transmit and reflective beamforming design for IRS-assisted multiuser MISO SWIPT systems," in *Proc. IEEE Int. Conf. Commun. (ICC)*, Jun. 2020, pp. 1–6.
- [75] H. Xie, J. Xu, and Y.-F. Liu, "Max-min fairness in IRS-aided multi-cell MISO systems with joint transmit and reflective beamforming," *IEEE Trans. Wireless Commun.*, vol. 20, no. 2, pp. 1379–1393, Feb. 2021.
- [76] C. Huang, Z. Yang, G. C. Alexandropoulos, K. Xiong, L. Wei, C. Yuen, Z. Zhang, and M. Debbah, "Multi-hop RIS-empowered terahertz communications: A DRL-based hybrid beamforming design," *IEEE J. Sel. Areas Commun.*, vol. 39, no. 6, pp. 1663–1677, Jun. 2021.
- [77] C. Huang, A. Zappone, G. C. Alexandropoulos, M. Debbah, and C. Yuen, "Reconfigurable intelligent surfaces for energy efficiency in wireless communication," *IEEE Trans. Wireless Commun.*, vol. 18, no. 8, pp. 4157–4170, Aug. 2019.
- [78] Q. Wu and R. Zhang, "Intelligent reflecting surface enhanced wireless network via joint active and passive beamforming," *IEEE Trans. Wireless Commun.*, vol. 18, no. 11, pp. 5394–5409, Nov. 2019.
- [79] J. Zhao, "A survey of intelligent reflecting surfaces (IRSs): Towards 6G wireless communication networks," 2019, *arXiv:1907.04789*.

2213

2214

2215

2216

2217

2218

2219

2220

2221

2222

2223

2224

2225

2226

2278

2279

2355

2356

2357

2358

2359

2360

2361

2362

2363

2364

2365

2366

2367

2368

2369

2370

2371

2372

2373

2374

2375

2376

2377

2378

2379

2380

2381

2382

2383

2384

2385

2386

2387

2388

2389

2390

2391

2392

2393

2394

2395

2396

2397

2398

2399

2400

2401

2402

2403

2404

2405

2406

2407

2408

2409

2410

2411

2412

2413

2414

2415

2416

2417

2418

2419

2420

2421

2422

2423

2424

2425

- [80] W. Tang, M. Z. Chen, J. Y. Dai, Y. Zeng, X. Zhao, S. Jin, Q. Cheng, and T. J. Cui, "Wireless communications with programmable metasurface: New paradigms, opportunities, and challenges on transceiver design," *IEEE Wireless Commun.*, vol. 27, no. 2, pp. 180–187, Apr. 2020.
- [81] S. A. Khaleefa, S. H. Alsamhi, and N. S. Rajput, "Tethered balloon technology for telecommunication, coverage and path loss," in *Proc. IEEE Students' Conf. Electr., Electron. Comput. Sci.*, Mar. 2014, pp. 1–4.
- [82] M. Samir, M. Elhattab, C. Assi, S. Sharafeddine, and A. Ghrayeb,
 "Optimizing age of information through aerial reconfigurable intelligent surfaces: A deep reinforcement learning approach," *IEEE Trans. Veh. Technol.*, vol. 70, no. 4, pp. 3978–3983, Apr. 2021.
- [83] L. Wang, K. Wang, C. Pan, and N. Aslam, "Joint trajectory and passive beamforming design for intelligent reflecting surface-aided UAV communications: A deep reinforcement learning approach," 2020, *arXiv:2007.08380*.
- [84] K. Guo, C. Wang, Z. Li, D. W. K. Ng, and K.-K. Wong, "Multiple UAV-borne IRS-aided millimeter wave multicast communications:
 A joint optimization framework," *IEEE Commun. Lett.*, vol. 25, no. 11, pp. 3674–3678, Nov. 2021.
- [85] H. Hashida, Y. Kawamoto, and N. Kato, "Intelligent reflecting surface placement optimization in air-ground communication networks toward 6G," *IEEE Wireless Commun.*, vol. 27, no. 6, pp. 146–151, Dec. 2020.
- [86] H. Lu, Y. Zeng, S. Jin, and R. Zhang, "Enabling panoramic full-angle reflection via aerial intelligent reflecting surface," in *Proc. IEEE Int. Conf. Commun. Workshops (ICC Workshops)*, Jun. 2020, pp. 1–6.
- [87] Y. Cai, Z. Wei, S. Hu, D. W. K. Ng, and J. Yuan, "Resource allocation for power-efficient IRS-assisted UAV communications," in *Proc. IEEE Int. Conf. Commun. Workshops (ICC Workshops)*, Jun. 2020, pp. 1–7.
- [88] Z. Wei, Y. Cai, Z. Sun, D. W. K. Ng, J. Yuan, M. Zhou, and L. Sun,
 "Sum-rate maximization for IRS-assisted UAV OFDMA communication systems," *IEEE Trans. Wireless Commun.*, vol. 20, no. 4, pp. 2530–2550,
 Apr. 2021.
- [89] Q.-U.-A. Nadeem, A. Kammoun, A. Chaaban, M. Debbah, and
 M.-S. Alouini, "Asymptotic max-min SINR analysis of reconfigurable
 intelligent surface assisted MISO systems," *IEEE Trans. Wireless Com- mun.*, vol. 19, no. 12, pp. 7748–7764, Dec. 2020.
- [90] C. You and R. Zhang, "3D trajectory optimization in Rician fading for
 UAV-enabled data harvesting," *IEEE Trans. Wireless Commun.*, vol. 18,
 no. 6, pp. 3192–3207, Jun. 2019.
- [91] Y. Sun, D. Xu, D. W. K. Ng, L. Dai, and R. Schober, "Optimal 3D-trajectory design and resource allocation for solar-powered UAV communication systems," *IEEE Trans. Commun.*, vol. 67, no. 6, pp. 4281–4298, Jun. 2019.
- Y. Cai, Z. Wei, R. Li, D. W. K. Ng, and J. Yuan, "Joint trajectory and resource allocation design for energy-efficient secure UAV communication systems," *IEEE Trans. Commun.*, vol. 68, no. 7, pp. 4536–4553, Jul. 2020.
- [93] D. Xu, Y. Sun, D. W. K. Ng, and R. Schober, "Multiuser MISO UAV communications in uncertain environments with no-fly zones: Robust trajectory and resource allocation design," *IEEE Trans. Commun.*, vol. 68, no. 5, pp. 3153–3172, May 2020.
- [94] B. Zheng and R. Zhang, "Intelligent reflecting surface-enhanced OFDM:
 Channel estimation and reflection optimization," *IEEE Wireless Commun. Lett.*, vol. 9, no. 4, pp. 518–522, Apr. 2020.
- A. Mahmoud, S. Muhaidat, P. C. Sofotasios, I. Abualhaol, O. A. Dobre, and H. Yanikomeroglu, "Intelligent reflecting surfaces assisted UAV communications for IoT networks: Performance analysis," *IEEE Trans. Green Commun. Netw.*, vol. 5, no. 3, pp. 1029–1040, Sep. 2021.
- [96] B. P. S. Sahoo, C.-C. Chou, C.-W. Weng, and H.-Y. Wei, "Enabling
 millimeter-wave 5G networks for massive IoT applications: A closer look
 at the issues impacting millimeter-waves in consumer devices under the
 5G framework," *IEEE Consum. Electron. Mag.*, vol. 8, no. 1, pp. 49–54,
 Jan. 2019.
- [97] X. Guan, Q. Wu, and R. Zhang, "Joint power control and passive beamforming in IRS-assisted spectrum sharing," *IEEE Commun. Lett.*, vol. 24, no. 7, pp. 1553–1557, Jul. 2020.
- I. Tang, X. Chen, X. Zhu, and F. Zhu, "Dynamic reallocation model of multiple unmanned aerial vehicle tasks in emergent adjustment scenarios," *IEEE Trans. Aerosp. Electron. Syst.*, vol. 59, no. 2, pp. 1139–1155, Apr. 2023.
- [99] J. Tang, G. Liu, and Q. Pan, "A review on representative swarm intelligence algorithms for solving optimization problems: Applications and trends," *IEEE/CAA J. Autom. Sinica*, vol. 8, no. 10, pp. 1627–1643, Oct. 2021.
 - VOLUME 11, 2023

- [100] C. Huang, G. Chen, P. Xiao, D. Mi, Y. Zhang, H. Tang, C. Lu, and R. Tafazolli, "Federated learning for RIS-assisted UAV-enabled wireless networks: Learning-based optimization for UAV trajectory, RIS phase shifts and weighted aggregation," in *Proc. 49th Annu. Conf. IEEE Ind. Electron. Soc. (IECON)*, Aug. 2023, pp. 1–7.
- [101] S. Niknam, H. S. Dhillon, and J. H. Reed, "Federated learning for wireless communications: Motivation, opportunities, and challenges," *IEEE Commun. Mag.*, vol. 58, no. 6, pp. 46–51, Jun. 2020.
- [102] Q.-V. Pham, M. Le, T. Huynh-The, Z. Han, and W.-J. Hwang, "Energyefficient federated learning over UAV-enabled wireless powered communications," *IEEE Trans. Veh. Technol.*, vol. 71, no. 5, pp. 4977–4990, May 2022.
- [103] D. Ma, L. Li, H. Ren, D. Wang, X. Li, and Z. Han, "Distributed rate optimization for intelligent reflecting surface with federated learning," in *Proc. IEEE Int. Conf. Commun. Workshops (ICC Workshops)*, Jun. 2020, pp. 1–6.
- [104] A. M. Elbir and S. Coleri, "Federated learning for channel estimation in conventional and RIS-assisted massive MIMO," *IEEE Trans. Wireless Commun.*, vol. 21, no. 6, pp. 4255–4268, Jun. 2022.
- [105] W. Saad, M. Bennis, and M. Chen, "A vision of 6G wireless systems: Applications, trends, technologies, and open research problems," *IEEE Netw.*, vol. 34, no. 3, pp. 134–142, May 2020.
- [106] Z. Peng, L. Li, M. Wang, Z. Zhang, Q. Liu, Y. Liu, and R. Liu, "An effective coverage scheme with passive-reflectors for urban millimeter-wave communication," *IEEE Antennas Wireless Propag. Lett.*, vol. 15, pp. 398–401, 2016.
- [107] T. Hong, J. Yao, C. Liu, and F. Qi, "MmWave measurement of RF reflectors for 5G green communications," *Wireless Commun. Mobile Comput.*, vol. 2018, May 2018, Art. no. 8217839.
- [108] E. Basar, M. Di Renzo, J. De Rosny, M. Debbah, M.-S. Alouini, and R. Zhang, "Wireless communications through reconfigurable intelligent surfaces," *IEEE Access*, vol. 7, pp. 116753–116773, 2019.
- [109] H. M. Hariz, S. Sheikhzadeh, N. Mokari, M. R. Javan, B. Abbasi-Arand, and E. A. Jorswieck, "AI-based radio resource management and trajectory design for PD-NOMA communication in IRS-UAV assisted networks," 2021, arXiv:2111.03869.
- [110] Y. M. Park, Y. K. Tun, Z. Han, and C. S. Hong, "Trajectory optimization and phase-shift design in IRS assisted UAV network for high speed trains," 2021, arXiv:2107.00857.
- [111] Y. Liu, X. Liu, X. Mu, T. Hou, J. Xu, M. Di Renzo, and N. Al-Dhahir, "Reconfigurable intelligent surfaces: Principles and opportunities," *IEEE Commun. Surveys Tuts.*, vol. 23, no. 3, pp. 1546–1577, 3rd Quart., 2021.
- [112] Z. Yang, M. Chen, W. Saad, C. S. Hong, and M. Shikh-Bahaei, "Energy efficient federated learning over wireless communication networks," *IEEE Trans. Wireless Commun.*, vol. 20, no. 3, pp. 1935–1949, Mar. 2021.
- [113] G. E. A. P. A. Batista, R. C. Prati, and M. C. Monard, "A study of the behavior of several methods for balancing machine learning training data," *ACM SIGKDD Explor. Newslett.*, vol. 6, no. 1, pp. 20–29, Jun. 2004.
- [114] C. O. Nnamani, M. R. A. Khandaker, and M. Sellathurai, "Joint beamforming and location optimization for secure data collection in wireless sensor networks with UAV-carried intelligent reflecting surface," 2021, arXiv:2101.06565.
- [115] H. Yang, J. Zhao, Z. Xiong, K.-Y. Lam, S. Sun, and L. Xiao, "Privacypreserving federated learning for UAV-enabled networks: Learning-based joint scheduling and resource management," *IEEE J. Sel. Areas Commun.*, vol. 39, no. 10, pp. 3144–3159, Oct. 2021.
- [116] C. Zhu, X. Zhu, J. Ren, and T. Qin, "Blockchain-enabled federated learning for UAV edge computing network: Issues and solutions," *IEEE Access*, vol. 10, pp. 56591–56610, 2022.
- [117] R. Myrzashova, S. H. Alsamhi, A. V. Shvetsov, A. Hawbani, and X. Wei, "Blockchain meets federated learning in healthcare: A systematic review with challenges and opportunities," *IEEE Internet Things J.*, vol. 10, no. 16, pp. 14418–14437, Aug. 2023.
- [118] B. Shang, L. Liu, J. Ma, and P. Fan, "Unmanned aerial vehicle meets vehicle-to-everything in secure communications," *IEEE Commun. Mag.*, vol. 57, no. 10, pp. 98–103, Oct. 2019.
- [119] H. Jiang, R. He, C. Ruan, J. Zhou, and D. Chang, "Three-dimensional geometry-based stochastic channel modeling for intelligent reflecting surface-assisted UAV MIMO communications," *IEEE Wireless Commun. Lett.*, vol. 10, no. 12, pp. 2727–2731, Dec. 2021.
- [120] C. Cao, Z. Lian, Y. Wang, Y. Su, and B. Jin, "A non-stationary geometrybased channel model for IRS-assisted UAV-MIMO channels," *IEEE Commun. Lett.*, vol. 25, no. 12, pp. 3760–3764, Dec. 2021.

- [121] B. McMahan, E. Moore, D. Ramage, S. Hampson, and B. A. Y. Arcas, 2427 "Communication-efficient learning of deep networks from decentralized 2428 data," in Proc. 20th Int. Conf. Artif. Intell. Statist., 2017, pp. 1273-1282. 2429
- 2430 [122] K. Bonawitz, H. Eichner, W. Grieskamp, D. Huba, A. Ingerman, V. Ivanov, C. Kiddon, J. Konecný, S. Mazzocchi, B. McMahan, 2431 2432 T. Van Overveldt, D. Petrou, D. Ramage, and J. Roselander, "Towards federated learning at scale: System design," in Proc. Mach. Learn. Syst., 2433 2434 vol. 1, 2019, pp. 374-388.
- [123] S. Samarakoon, M. Bennis, W. Saad, and M. Debbah, "Distributed fed-2435 2436 erated learning for ultra-reliable low-latency vehicular communications," IEEE Trans. Commun., vol. 68, no. 2, pp. 1146-1159, Feb. 2020. 2437
- [124] S. Wang, M. Chen, C. Yin, W. Saad, C. S. Hong, S. Cui, and H. V. Poor, 2438 "Federated learning for task and resource allocation in wireless high-2439 2440 altitude balloon networks," IEEE Internet Things J., vol. 8, no. 24, pp. 17460-17475, Dec. 2021. 2441
- 2442 [125] J. S. Ng, W. Y. B. Lim, H.-N. Dai, Z. Xiong, J. Huang, D. Niyato, X.-S. Hua, C. Leung, and C. Miao, "Joint auction-coalition formation 2443 2444 framework for communication-efficient federated learning in UAV-enabled 2445 Internet of Vehicles," IEEE Trans. Intell. Transp. Syst., vol. 22, no. 4, pp. 2326-2344, Apr. 2021. 2446
- 2447 [126] H. Shiri, J. Park, and M. Bennis, "Communication-efficient massive UAV online path control: Federated learning meets mean-field game theory," 2448 IEEE Trans. Commun., vol. 68, no. 11, pp. 6840-6857, Nov. 2020. 2449
- [127] N. I. Mowla, N. H. Tran, I. Doh, and K. Chae, "Federated learning-based 2450 cognitive detection of jamming attack in flying ad-hoc network," IEEE 2451 2452 Access, vol. 8, pp. 4338-4350, 2020.
- [128] N. I. Mowla, N. H. Tran, I. Doh, and K. Chae, "AFRL: Adaptive federated 2453 reinforcement learning for intelligent jamming defense in FANET," 2454 J. Commun. Netw., vol. 22, no. 3, pp. 244-258, Jun. 2020. 2455
- 2456 [129] Z. Md. Fadlullah and N. Kato, "HCP: Heterogeneous computing platform for federated learning based collaborative content caching 2457 towards 6G networks," IEEE Trans. Emerg. Topics Comput., vol. 10, no. 1, 2458 pp. 112-123, Jan. 2022. 2459
- [130] Y. Liu, J. Nie, X. Li, S. H. Ahmed, W. Y. B. Lim, and C. Miao, "Federated 2460 learning in the sky: Aerial-ground air quality sensing framework with 2461 UAV swarms," IEEE Internet Things J., vol. 8, no. 12, pp. 9827-9837, 2462 2463 Jun. 2021.
- [131] W. Y. B. Lim, J. Huang, Z. Xiong, J. Kang, D. Niyato, X.-S. Hua, 2464 C. Leung, and C. Miao, "Towards federated learning in UAV-enabled 2465 Internet of Vehicles: A multi-dimensional contract-matching approach," 2466 2467 IEEE Trans. Intell. Transp. Syst., vol. 22, no. 8, pp. 5140-5154, 2468 Aug. 2021.
- [132] K. Bonawitz, V. Ivanov, B. Kreuter, A. Marcedone, H. B. McMahan, 2469 S. Patel, D. Ramage, A. Segal, and K. Seth, "Practical secure aggregation 2470 for federated learning on user-held data," 2016, arXiv:1611.04482. 2471
- 2472 [133] H. Kim, J. Park, M. Bennis, and S.-L. Kim, "Blockchained on-device federated learning," IEEE Commun. Lett., vol. 24, no. 6, pp. 1279-1283, 2473 2474 Jun. 2020.
- [134] Y. Qu, L. Gao, T. H. Luan, Y. Xiang, S. Yu, B. Li, and G. Zheng, 2475 2476 "Decentralized privacy using blockchain-enabled federated learning in 2477 fog computing," IEEE Internet Things J., vol. 7, no. 6, pp. 5171-5183, 2478 Jun 2020
- [135] Y. Lu, X. Huang, Y. Dai, S. Maharjan, and Y. Zhang, "Blockchain 2479 2480 and federated learning for privacy-preserved data sharing in industrial IoT," IEEE Trans. Ind. Informat., vol. 16, no. 6, pp. 4177-4186, 2481 Jun. 2020. 2482
- [136] Y. J. Kim and C. S. Hong, "Blockchain-based node-aware dynamic 2483 2484 weighting methods for improving federated learning performance," in Proc. 20th Asia-Pacific Netw. Operations Manage. Symp. (APNOMS), 2485 Sep. 2019, pp. 1-4. 2486
- [137] Y. Wang, Z. Su, N. Zhang, and A. Benslimane, "Learning in the air: 2487 Secure federated learning for UAV-assisted crowdsensing," IEEE Trans. 2488 2489 Netw. Sci. Eng., vol. 8, no. 2, pp. 1055-1069, Apr. 2021.
- [138] J. Konečný, H. B. McMahan, F. X. Yu, P. Richtárik, A. T. Suresh, and 2490 2491 D. Bacon, "Federated learning: Strategies for improving communication efficiency," 2016, arXiv:1610.05492. 2492
- 2493 [139] L. Wang, W. Wang, and B. Li, "CMFL: Mitigating communication overhead for federated learning," in Proc. IEEE 39th Int. Conf. Distrib. 2494 Comput. Syst. (ICDCS), Jul. 2019, pp. 954-964. 2495
- 2496 [140] S. Li, S. M. M. Kalan, A. S. Avestimehr, and M. Soltanolkotabi, "Near-optimal straggler mitigation for distributed gradient methods," in 2497 2498 Proc. IEEE Int. Parallel Distrib. Process. Symp. Workshops (IPDPSW), 2499 May 2018, pp. 857-866.

- [141] Q. Wu, S. Zhang, B. Zheng, C. You, and R. Zhang, "Intelligent 2500 reflecting surface-aided wireless communications: A tutorial," IEEE 2501 Trans. Commun., vol. 69, no. 5, pp. 3313-3351, May 2021. 2502
- [142] Q. Wu, J. Xu, Y. Zeng, D. W. K. Ng, N. Al-Dhahir, R. Schober, and A. L. Swindlehurst, "A comprehensive overview on 5G-and-beyond networks with UAVs: From communications to sensing and intelligence," IEEE J. Sel. Areas Commun., vol. 39, no. 10, pp. 2912–2945, Oct. 2021.
- [143] S. Alfattani, W. Jaafar, Y. Hmamouche, H. Yanikomeroglu, A. Yongaçoglu, N. D. Dào, and P. Zhu, "Aerial platforms with 2508 reconfigurable smart surfaces for 5G and beyond," IEEE Commun. 2509 Mag., vol. 59, no. 1, pp. 96-102, Jan. 2021.
- [144] L. Gupta, R. Jain, and G. Vaszkun, "Survey of important issues in UAV communication networks," IEEE Commun. Surveys Tuts., vol. 18, no. 2, pp. 1123-1152, 2nd Quart., 2016.
- [145] B. Duo, Q. Wu, X. Yuan, and R. Zhang, "Energy efficiency maximization for full-duplex UAV secrecy communication," IEEE Trans. Veh. Technol., vol. 69, no. 4, pp. 4590-4595, Apr. 2020.
- [146] D. W. Matolak and R. Sun, "Air-ground channel characterization for unmanned aircraft systems-Part III: The suburban and near-urban environments," IEEE Trans. Veh. Technol., vol. 66, no. 8, pp. 6607-6618, Aug. 2017.
- [147] Y. Liu, Z. Qin, Y. Cai, Y. Gao, G. Y. Li, and A. Nallanathan, "UAV communications based on non-orthogonal multiple access," IEEE Wireless Commun., vol. 26, no. 1, pp. 52-57, Feb. 2019.
- [148] Y. Liu, Z. Qin, M. Elkashlan, Z. Ding, A. Nallanathan, and L. Hanzo, "Nonorthogonal multiple access for 5G and beyond," Proc. IEEE, vol. 105, no. 12, pp. 2347-2381, Dec. 2017.
- [149] B. Zheng, Q. Wu, and R. Zhang, "Intelligent reflecting surface-assisted multiple access with user pairing: NOMA or OMA?" IEEE Commun. Lett., vol. 24, no. 4, pp. 753-757, Apr. 2020.
- [150] X. Mu, Y. Liu, L. Guo, J. Lin, and H. V. Poor, "Intelligent reflecting surface enhanced multi-UAV NOMA networks," *IEEE J. Sel. Areas* Commun., vol. 39, no. 10, pp. 3051-3066, Oct. 2021.
- [151] C. You and R. Zhang, "Hybrid offline-online design for UAV-enabled data harvesting in probabilistic LoS channels," IEEE Trans. Wireless Commun., vol. 19, no. 6, pp. 3753-3768, Jun. 2020.
- [152] B. Li, Z. Fei, and Y. Zhang, "UAV communications for 5G and beyond: Recent advances and future trends," IEEE Internet Things J., vol. 6, no. 2, pp. 2241–2263, Apr. 2019.
- [153] L. Zheng and D. N. C. Tse, "Diversity and multiplexing: A fundamental 2539 tradeoff in multiple-antenna channels," IEEE Trans. Inf. Theory, vol. 49, 2540 no. 5, pp. 1073-1096, May 2003. 2541
- [154] H. Li, J. Xi, M. He, and X. Wang, "Beam tracking method based on reconfigurable intelligent surface for obstructed communication," Chin. J. Aeronaut., vol. 35, no. 8, pp. 158-167, Aug. 2022.
- [155] H. Song, L. Liu, B. Shang, S. Pudlewski, and E. S. Bentley, "Enhanced flooding-based routing protocol for swarm UAV networks: Random network coding meets clustering," in Proc. IEEE Conf. Comput. Commun. (INFOCOM), May 2021, pp. 1-10.



ALEXEY V. SHVETSOV received the Ph.D. degree from the Russian University of Transport, Moscow, Russia, in 2018. He is currently an Associate Professor with the Department of Smart Technologies, Moscow Polytechnic University, and an Associate Professor with the Department of Operation of Road Transport and Car Service, North-Eastern Federal University. He has authored or coauthored over 100 publications in refereed journals and conferences. His current research

interest includes smart technologies.

2533

2534

2535

2536

2537

2538

2542

2543

2544

2545

2546

2547

2548

2521

2503

2504

2505

2506

2507

2510

2511

2512

2513

2514

2515



SAEED HAMOOD ALSAMHI received the B.Eng. degree from the Department of Electronic Engineering (Communication Division), Ibb University, Yemen, in 2009, and the M.Tech. degree in communication systems and the Ph.D. degree from the Department of Electronics Engineering, Indian Institute of Technology (Banaras Hindu University) Varanasi (IIT BHU) Varanasi, Varanasi, India, in 2012 and 2015, respectively. In 2009, he was a Lecturer Assistant with the Engineering

Faculty, Ibb University. After that, he held a postdoctoral position with the School of Aerospace Engineering, Tsinghua University, Beijing, China, in optimal and smart wireless network research and its applications to enhance robotics technologies. Since 2019, he has been an Assistant Professor with the Shenzhen Institutes of Advanced Technology, Chinese Academy of Sciences, Shenzhen. In 2020, he was a MSCA Smart 4.0 Fellow with the Athlone Institute of Technology, Athlone, Ireland. He has published more than 120 articles in high-reputation journals in IEEE, Elsevier, Springer, Wiley, and MDPI publishers. His research interests include green and semantic communication, the green Internet of Things, QoE, QoS, multi-robot collaboration, blockchain technology, federated learning, and space technologies (high altitude platforms, drones, and tethered balloon technologies).



AMMAR HAWBANI received the B.S., M.S., and Ph.D. degrees in computer software and theory from the University of Science and Technology of China (USTC), Hefei, China, in 2009, 2012, and 2016, respectively. He is currently an Associate Professor in networking and communication algorithms with the School of Computer Science and Technology, USTC, where he was a Postdoctoral Researcher with the School of Computer Science and Technology, from 2016 to 2019. His research

interests include the IoT, WSNs, WBANs, WMNs, VANETs, and SDN.



SANTOSH KUMAR (Member, IEEE) is currently an Assistant Professor in computer science and engineering. Prior to joining IIIT-NR, he was a Ph.D. Scholar with the Department of Computer Science and Engineering, Indian Institute of Technology (Banaras Hindu University) Varanasi (IIT BHU) Varanasi, Varanasi, Uttar Pradesh, India. He is an active member of the Computer Society and the Association for Computing Machinery. He has published over 14 journals (SCI-index

journal) and five conference papers, including two tier-1 international conferences (2016 ACM Multimedia and 2016 Mobisys); filed one patent, and 14 book chapters in edited books (Springer publication and IGI publication).



SUMIT SRIVASTAVA (Graduate Student Member, IEEE) received the M.Tech. degree in digital communication from UKTU Dehradun. He is currently pursuing the Ph.D. degree with the Indian Institute of Technology (Banaras Hindu University) Varanasi, Varanasi. He has more than 15 years of teaching experience. He is also an Assistant Professor (Senior Scale) with the Department of Electronics and Communication Engineering, FET, MJP Rohilkhand University

Bareilly, India. His research interests include RADAR systems, microstrip patch antennas, the IoT, artificial intelligence, machine learning, and gas sensors.



SWETA AGARWAL received the B.Tech. and M.Tech. degrees in electronics and communication engineering and microwave engineering from Uttar Pradesh Technical University (Dr. APJ Abdul Kalam Technical University), Lucknow. She is currently pursuing the Ph.D. degree with the Indian Institute of Technology (Banaras Hindu University) Varanasi, Varanasi. She has more than 15 years of teaching experience. She is also an Assistant Professor with the Department of

Electronics and Communication Engineering, Invertis University, Bareilly. Her research interests include antenna designing and microwave engineering, the IoT, artificial intelligence, and machine learning.



NAVIN SINCH RAJPUT (Member, IEEE) received the M.E. degree in communication systems, in 1997, and the Ph.D. degree in the area of intelligent data analysis and pattern recognition techniques from the Indian Institute of Technology (Banaras Hindu University) Varanasi (IIT BHU) Varanasi, Varanasi, Uttar Pradesh, in 2011. He is currently an Associate Professor with the Department of Electronics Engineering, IIT (BHU) Varanasi. His research interests include

intelligent gas sensors and systems, embedded and real-time artificial intelligence-based systems, cyber-physical systems, and digital twins for smart cities. He has authored books and e-content on smart systems and over 60 publications in refereed journals and conferences. He is also developing intelligent air quality monitoring systems for widespread commercialization at national and international scales as a spin-off company.



AMR A. ALAMMARI received the B.Tech. degree in electronics, communication from Ibb University, Ibb, Yemen, the M.Tech. degree in electronics and communication from Jawaharlal Nehru Technological University, India, and the Ph.D. degree in wireless communications from AMU, India. He is currently an Assistant Professor with the Department of Electrical Engineering, Ibb University. He is also a Lecturer with National University, Yemen, and International Malaysian

University, Yemen. He has published many research papers in the area of wireless communication. His research interests include wireless communication and networking technologies with a focus on 5G/6G wireless systems.



FARHAN M. A. NASHWAN received the Master of Science degree in electrical engineering from the Electronic and Communication Engineering Department, Faculty of Engineering, Al-Mustansiriyah University, Baghdad, Iraq, and the Ph.D. degree in electronics and electrical communications engineering from Cairo University, Egypt, in 2014. He is currently an Assistant Professor and the Dean of the Faculty of Engineering, Ibb University, Ibb, Yemen. His current

research interests include clustering, object tracking, pattern recognition, and computer vision.